# **COLLECTIVE AGREEMENT NO. 1**

between

UNIFOR

and

VIA RAIL CANADA INC.

covering

**OFF-TRAIN EMPLOYEES** 

2022-2023-2024

## EMPLOYEE ASSISTANCE PROGRAM 1-844-880-9142 (English)

PROGRAMME D'ASSISTANCE AUX EMPLOYÉS 1-844-880-9143 (Français)

<b>ARTICLE</b>		<u>PAGE</u>
ARTICLE 1	DEFINITIONS	1
ARTICLE 2	RECOGNITION AND SCOPE	2
ARTICLE 3	DEDUCTION OF UNION DUES	4
ARTICLE 4	Hours of Work	6
ARTICLE 5	OVERTIME AND CALLS	10
ARTICLE 6	REST DAYS	12
ARTICLE 7	Spare Boards	13
ARTICLE 8	GENERAL HOLIDAYS	14
ARTICLE 9	VACATIONS	16
ARTICLE 10	SENIORITY GROUPINGS	21
ARTICLE 11	SENIORITY	22
ARTICLE 12	BULLETINING AND FILLING OF POSITIONS	28
ARTICLE 13	STAFF REDUCTION, DISPLACEMENTAND RECALL TO SERVICE	33
ARTICLE 14	TRANSFERS	
ARTICLE 15	REHABILITATION	36.
ARTICLE 16	TRAINING	
ARTICLE 17	LEAVE OF ABSENCE & FREE TRANSPORTATION	40
ARTICLE 18	SERVICE AWAY FROM HOME HEADQUARTERS	41
ARTICLE 19	Attending Court	42
ARTICLE 20	Held for Investigation or Company Business	42
ARTICLE 21	RELIEF WORK AND PRESERVATION OF RATES	42
ARTICLE 22	Service Letters	43
ARTICLE 23	RATES OF PAY	44
ARTICLE 24	DISCIPLINE AND GRIEVANCE PROCEDURE	45
ARTICLE 25	FINAL SETTLEMENT OF DISPUTES	47
ARTICLE 26	HEALTH AND WELFARE	47
ARTICLE 27	GENERAL	48
ARTICLE 28	WAGE RATES FOR NEW JOBS	52
ARTICLE 29	Bereavement Leave	53

# INDEX

<u>ARTICLE</u>	Ē	PAGE
ARTICLE 30	JOB SECURITY, TECHNOLOGICAL, OPERATIONAL AND ORGANIZATIONAL CHANGES	54
ARTICLE 31	JURY DUTY	55
ARTICLE 32	SHIFT DIFFERENTIALS	55
ARTICLE 33	PENSION PLAN WIND UP	56
ARTICLE 34	LIFE INSURANCE UPON RETIREMENT	56
ARTICLE 35	DENTAL PLAN	56
ARTICLE 36	EXTENDED HEALTH CARE PLAN	56
ARTICLE 37	PAID MATERNITY LEAVE	56
ARTICLE 38	UNIFORMS	57
ARTICLE 39	SAFETY, HEALTH AND ENVIRONMENT	58
ARTICLE 40	DURATION OF COLLECTIVE AGREEMENT	59

<u>APPENDIX</u>		PAGE
APPENDIX A	WAGE SCALE	61
APPENDIX B	REGULATIONS GOVERNING THE APPLICATION OFSUPPLEMENTAL SENIORITY LIST	rs68
APPENDIX C	LETTER ON CONTRACTING OUT	70
APPENDIX D	BILINGUALISM	72
APPENDIX E	LETTER RE PART-TIME EMPLOYEES	74
APPENDIX F	LETTER ON PAYMENT DURING TRAINING	75
APPENDIX G	LEFT BLANK BY THE PARTIES	76
APPENDIX H	LETTER ON SPLITTING OF "BEREAVEMENT LEAVE"	79
APPENDIX I	LETTER ON PAYMENT FOR SICK LEAVE	
APPENDIX J	LETTER CONCERNING THE ASSESSMENT OF DISCIPLINE FOR MINOR OFFENCES WITHOUT THE NECESSITY OF HOLDING AN INVESTIGATION UNDER ARTICLE 24.2	82
APPENDIX K	LETTER CONCERNING EMPLOYEES IN COLLECTIVE AGREEMENT NO. 2 WHO ARE MEDICALLY UNFIT TO PERFORM THEIR REGULAR DUTIES DUE TO PREGNANCY	84
APPENDIX L	INTENTIONALLY LEFT BLANK BY THE PARTIES	
APPENDIX M	LETTER CONCERNING THE INTERPRETATION AND APPLICATION OF THE WORD "TERMINAL", AS USED THROUGHOUT COLLECTIVE AGREEMENT NO.1	7
APPENDIX N	INTENTIONALLY LEFT BLANK BY THE PARTIES	
APPENDIX O	LETTER CONCERNING THE BREAK PERIODS FOR THE EMPLOYEES	
	Working in the Telephone Sales Office	90
APPENDIX P	LETTER CONCERNING THE APPLICATION OF ARTICLE 4.26	92
APPENDIX Q	LETTER CONCERNING THE APPLICATION OF ARTICLE 11.9	93
APPENDIX R	SKILL SETS	
APPENDIX S	LETTER OF UNDERSTANDING LOCOMOTIVE ENGINEER TRAINING-CLOSED PE COMMITMENT	
APPENDIX T	LETTER CONCERNING RECOGNITION PROGRAM	97
APPENDIX U	LETTER CONCERNING MATERNITY-RELATED REASSIGNED AND LEAVE	98
APPENDIX V	ARTICLES 4 AND 4.11 COLLECTIVE AGREEMENT #1	100
APPENDIX W	RED-CIRCLED FULL-TIME EMPLOYEES	101
APPENDIX X	CANADIAN ARMED FORCES RESERVES AT VIA RAIL	102
APPENDIX Y	LETTER CONCERNING CONSENT FOR UNREDUCED PENSION	103

## LETTER OF UNDERSTANDING

LANGUAGE TRAINING	.112
VACATIONS SOUTH WESTERN ONTARIO	.113
LETTER ON SHIFT TRADES	.114
RATE OF PAY OF MONTREAL STATION IN-CHARGE	.115
JOB SHARING	.116
ARTICLE 38.6 OF THE COLLECTIVE AGREEMENT	.120
COLLECTIVE AGREEMENT SEND BY EMAIL	.121
ONGOING LABOUR RELATIONS COMMITTEE	.122
EDUCATIONAL LEAVE	.124
TRAINING/DIFFICULT CUSTOMER INTERACTIONS	.126
RECLASSIFY SENIOR TELEPHONE AGENT IN MONCTON AND MONTREAL	.127
WOMEN'S ADVOCACY Program at VIA Rail	128

## **ARTICLE 1**

#### DEFINITIONS

#### 1.1 Work Week

For regularly assigned employees - a week beginning on the first day on which the assignment is bulletined to work.

#### 1.2 Employee

The word "employee" as used hereinafter shall be understood to mean any employee holding seniority under this Agreement.

#### 1.3 Temporary Vacancy

A vacancy in a position caused by the regularly assigned occupant being absent from duty (including on vacation but excluding pre-retirement vacation) or temporarily assigned to other duties.

#### 1.4 Clerk

The classification Clerk may be subdivided as follows:

Clerical Worker - Employees who regularly devote not less than four hours per day to the writing and calculating incident to keeping records and accounts, rendition of bills, reports and statements, handling of correspondence and similar work.

Machine Operator - Employees who regularly devote not less than four hours per day to the operation of office or station mechanical equipment requiring special skill and training, such as accounting, calculating, statistical and key punch machines, typewriters, dictaphone, telephones and other similar equipment.

The foregoing shall not be construed to apply to:

- Employees engaged in sorting bills, inserting and removing carbons, etc.;
- Junior Clerks, or to other employees doing similar work;
- Employees performing manual work not requiring clerical ability.

### 1.5 Mutually Arranged (or Mutually Agreed)

An agreement in writing between the proper officer of the Corporation and the designated National or Regional Representative of the Union.

#### 1.6 Locally Arranged

An agreement between the local supervisory officer of the Corporation and the Local Chairperson of the Union.

**1.7** The use of the masculine gender in this collective agreement includes the feminine and vice versa.

# 1.8 A years' service is defined as 250 days cumulative compensated service.

## ARTICLE 2 RECOGNITION AND SCOPE

- 2.1 The Corporation recognizes UNIFOR as the sole collective bargaining agent with respect to wages, hours of work and other working conditions for all classes of employees recognized by the Canada Labour Relations Board certification order dated January 25, 1985, as well as equipment maintenance employees in the classifications represented by the Union.
- **2.2** In-charge positions will be created where required, in accordance with the established selection process:
  - (a) Employees will be subject to an evaluation process to establish their suitability for the position prior to training in seniority order.
  - (b) Employees will be provided with job aids to assist them in the selection process.
  - (c) If an interview is part of the selection process, the employee may have a union representative present as an observer during the interview. Whenever practicable the employee's supervisor should not be one of the interviewers.
  - (d) Employees will be advised of the passing grade(s) required before undergoing the selection process. They will be provided with their grade(s) at the end of the process and have an opportunity for feedback as to their performance.

(e) Employees must subsequently successfully complete the required training program.

Once employees are qualified for the in-charge position, seniority will determine their job selection.

The in-charge position will be paid a premium rate of pay of \$50 per week above the rate of pay of the senior position within their work group.

The in-charge position will be designated bilingual in locations where the Corporation deems it is an operational requirement.

- 2.3 It is the policy of the Corporation to cooperate in every practical way with employees who desire advancement to official or excepted positions. Accordingly, such employees who make application to their supervisor or the Manager, Human Resources stating their desires, qualifications and experience will be given preference for openings in classifications in excepted positions providing they have the necessary capabilities.
- 2.4 The Corporation also accepts that the main function of Supervisors is to direct the work force. Supervisors and employees outside this bargaining unit should not engage, normally, in work currently or traditionally performed by employees in this bargaining unit.
- 2.5 Should there be a "sale of business" within the context of the Canada Labour Code, the provisions of this collective agreement shall be binding upon any successor or merged company or companies or any successor in the control of the Corporation. In the event there is a merger with another company in which the bargaining unit employees therein are represented by another union in such company, the representation rights and status quo of the Union shall be maintained until a final determination is made under the Canada Labour Code as to the proper representative of the combined group.
- 2.6 The Corporation shall acquaint all new employees, or employees who transfer from another agreement to work under this agreement, with the fact that a collective agreement is in effect. The Corporation shall supply a copy of this agreement and the Supplemental Agreement to all employees. Whenever possible, the Corporation will meet with the new employee within the first 10 working days to acquaint the employee with the conditions of employment, the Collective Agreement and the benefit package during which time the Local Chairperson, or authorized committee person will be given the opportunity to make a brief presentation of 30 minutes on behalf of the Union.

2.7 When the Corporation sells, leases, merges, amalgamates or transfers or agrees to sell, lease, merge, amalgamate or transfer its business or the operations thereof or any part of either of them, the Corporation or the purchaser, lessee or transferee or any of them will be a party to and be bound by the existing Collective Agreement/agreements or subsequent Collective Agreement/agreements entered into with the bargaining agent representing any employees affected by the sale, lease, merger, amalgamation, transfer or contract.

The Collective Agreement continues in force and is binding upon the parties to the aforementioned conditions.

#### ARTICLE 3 DEDUCTION OF UNION DUES

#### 3.1

The Corporation shall deduct a set percentage from wages due and payable to each employee coming within the scope of this Collective Agreement. The percentage shall be 1.355% and payable every pay period, subject to the conditions and exceptions set forth hereunder and shall not include initiation fees or special assessments. The amount to be deducted shall not be changed during the term of this Agreement excepting to conform with a change in the amount of regular dues in accordance with the Union's constitutional provisions *and* shall be applicable on receipt by the Corporation of notice in writing from the Union of the change in percentage to deduct.

- **3.2** Intentionally left blank by the parties
- **3.3** Employees filling positions of a supervisory or confidential nature not subject to all the rules of the agreement as may be mutually agreed between the designated officers of the Corporation and of the Union shall be excepted from dues deductions.
- **3.4** Membership in the Union signatory hereto shall be available to any employee eligible under the constitution of the Union on payment of the initiation or reinstatement fees uniformly required of all other such applicants by the local concerned. Membership shall not be denied for reasons of race, national origin, color, religion, or gender.
- **3.5** Deductions for new employees shall commence on the first pay period.
- **3.6** Employees filling positions in more than one wage agreement during a pay period will pay union dues under the wage agreement in which they are working on the day the dues are deducted.

- **3.7** Only payroll deductions now or hereafter required by law, deduction of monies due or owing the Corporation and pension deductions shall be made from wages prior to the deduction of dues.
- **3.8** The amounts of dues so deducted from wages accompanied by a statement of deductions from individuals shall be remitted by the Corporation to the officer or officers of the Union, as may be mutually agreed by the Corporation and the Union, not later than 40 calendar days following the pay period in which the deductions are made.
- **3.9** The Corporation shall not be responsible financially or otherwise, either to the Union or to any employee, for any failure to make deductions or for making improper or inaccurate deductions or remittances. However, in any instance in which an error occurs in the amount of any deduction of dues from an employee's wages, the Corporation shall adjust it directly with the employee. In the event of any mistake by the Corporation in the amount of its remittances to the Union, the Corporation shall adjust the amount in a subsequent remittance. The Corporation's liability for any and all amounts deducted pursuant to the provisions of this Article shall terminate at the time it remits the amounts payable to the designated officer or officers of the Union.
- **3.10** The question of what, if any, compensation shall be paid to the Corporation by the Union in recognition of services performed under this Article shall be left in abeyance subject to reconsideration at the request of either party on 15 days' notice in writing.
- **3.11** In the event of any action at law against the parties hereto or either of them resulting from any deduction or deductions from payrolls made or to be made by the Corporation pursuant to Article 3.1, both parties shall cooperate fully in the defense of such action. Each party shall bear its own cost of such defense except that if at the request of the Union counsel fees are incurred these shall be borne by the Union. Save as aforesaid the Union shall indemnify and save harmless the Corporation from any losses, damages, cost, liability, or expenses suffered or sustained by it as a result of any such deduction or deductions from payrolls.
- **3.12** The Corporation will pay monthly to the Union \$0.13 per compensated hour for each employee in the bargaining unit to reimburse Union Representatives for time off the job in the performance of union duties on behalf of employees within bargaining units at VIA represented by the Union.

## ARTICLE 4 HOURS OF WORK

#### Articles 4.12 to 4.28 are applicable to part-timework.

**4.1** Except as otherwise provided in Articles 4.2, 4.3, and 4.5 and in the Wage Scale, eight consecutive hours of service, exclusive of the meal period, shall constitute a day's work.

**Note:** refer to Appendix V.

- **4.2** Employees may be assigned to work eight consecutive hours and allowed 30 minutes in which to eat without deduction in pay, said meal period to begin no sooner than the end of the third hour and end no later than the beginning of the fifth hour.
- **4.3** Where the work is of an intermittent character, there being no work for periods of more than one hour's duration for one or more employees, and their services cannot otherwise be utilized, split trick assignments may be established. Such split trick assignments will be confined to not more than two tours of duty within a spread of twelve consecutive hours. Split trick assignments will not be established until agreed to by the designated National or Regional Representative of the Union or, in the case of a dispute, until a decision has been rendered authorizing establishment. The spread of hours may be extended by mutual agreement to take care of exception conditions.
- 4.4 Where it has been the practice for weekly rated employees to work less than eight hours per day, that practice shall be continued unless changed on account of conditions beyond the control of the Corporation. Should conditions occasionally demand, employees working such reduced hours may be required to work eight hours per day and overtime will not accrue until after eight hours' service has been performed. To take care of regular requirements such employees may be required to work extra hours on certain days and overtime shall only accrue after eight hours' service has been performed.
- **4.5** Regularly assigned employees who report for duty on their regular assignments shall be paid eight hours at their regular rate. Employees who are permitted to leave work at their own request shall be paid at the hourly rate for actual time worked, except as may be otherwise arranged locally.

An employee prevented from completing a shift due to a bona fide injury sustained while on duty will be paid for his full shift at straight time rates of pay, unless the employee receives Worker's Compensation benefits for the day of the injury in which case the employee will be paid the difference between such compensation and payment for the full shift.

- **4.6** Employees shall be allowed a regular meal period of not less than 30 minutes nor more than one hour, said meal period to begin no sooner than the end of the third hour, and end no later than the beginning of the fifth hour. Should an employee not be allowed a meal period within the agreed hours, he shall be paid for his meal period at the punitive rate and at the first opportunity allowed 30 minutes for lunch without deduction in pay.
- **4.7** The starting time of employees on regular full-time assignments shall not vary by more than two hours from the earliest to the latest starting time on all days of the week. Not less than seven calendar days' notice will be given when changes are required unless circumstances justify a shorter notice. The employee and the local chairperson shall be notified in writing of such changes. Regular relief assignments will correspond to the starting time, duties and work locations of the employees relieved.
- **4.8** Unless necessary to meet the requirements of the service, employees will not be required to commence work between 0001 hours and 0600 hours. Reasons for such requirements will be provided to the designated National or Regional Representative of the Union in writing not less than seven calendar days prior to the commencement of such shift.
- **4.9** Regular relief assignments will provide 80 hours of work and four rest days over a 14-day period. Whenever possible, such assignments will be for five days' work per week and two rest days, preferably consecutive. Regular relief assignments will in all cases provide at least one rest day per calendar week; however, when an employee bids different successive relief assignments, that condition may not be fulfilled. They shall be established to perform necessary relief work or to perform relief work on certain days and such types of other work on other days as may be assigned under this Agreement.
- **4.10** Where it is impracticable to establish relief assignments in accordance with Article 4.9, the designated National or Regional Representative of the Union and the proper officer of the Corporation may by mutual agreement arrange for relief assignments on such other basis as may be suitable. Consent to such proposed arrangements shall not be unreasonably withheld in cases where employees would otherwise be required to work on assigned rest days or unreasonable travel time would be involved.
- **4.11** Notwithstanding the provisions of Articles 4.5 and 6, regular assignments consisting of four days of 10 hours may be established as mutually arranged.

Note: refer to Appendix V.

- **4.12** The Corporation recognizes that every effort shall be made to establish and maintain as many regular full-time assigned positions as is practical. Within that context, part-time employees can be defined as employees required to perform work during certain periods of a day or on days of the week which, taking into account staffing requirements, cannot practically be made part of regular full time assigned positions in accordance with the applicable provisions of the Collective Agreement. The Corporation will review any alternative put forth by the Union to establish the feasibility of creating regular full-time assignments.
- **4.13** The maximum number of part-time employees at any point in time will not exceed one such employee for every three regular full-time assigned positions established system wide. The Corporation will supply to one designated Union officer the number of regular full-time positions and regular part-time assignments in the system. This report will be supplied to one designated Union officer on a quarterly basis thereafter.
  - **NOTE:** For the purposes of calculating the one for three ratios, the number of part-time employees relieving on regular full-time assigned positions will not be included in the calculation. Reductions of the labour force causing employees displaced from regular full-time assignments to elect to protect part-time work will not be construed as reason to exceed the maximum number of part-time employees permitted at any given time.

The junior part-time employee will be placed on laid-off status if an employee displaced in accordance with Article 13.3 elects to protect part-time work when the Corporation already has the maximum number of part-time employees permitted. Senior laid-off employees may be recalled at their option to part-time work to fill out the number of part-time employees permitted at any point in time. The monitoring of the part-time ratio will be at the System level and any grievances that result there from will take place at Step 3 of the grievance procedure.

Where regular part-time assignments have been established as mutually arranged, in accordance with the provisions of Articles 4.12 through 4.26 and where the Corporation determines that a part time assignment is no longer required, a senior part-time employee holding such assignment may displace a junior part-time employee in his region for whose position he is qualified.

In the event a senior part-time employee displaces another part-time employee outside their seniority group (JSET), the conditions of article 27.5 will apply.

4.14 Part-time employees will be called in seniority order (unless otherwise locally arranged) but will not be required to work more than one tour of duty in any calendar day, though they may elect to accept a second tour of duty on a voluntary basis. Such tours of duty to be paid at regular rates.

- **4.15** Part-time employees shall not be required to report for work unless provided with a minimum of three hours' notice prior to the commencement time of the work assignment required. The calling procedure will be as locally arranged.
- **4.16** If the senior part-time employee cannot be contacted or fails to respond, the guarantee referred to in Article 4.25 will be reduced by the number of hours the employee would have worked. The junior employee must accept the call unless there is a bona fide reason.
- **4.17** Part-time employees will be guaranteed a minimum of four hours' pay for each tour of duty, except upon relieving a regular full-time assignment, in which case the part-time employee shall, subject to Article 4.5 or 4.11, be paid the regular hours of the shift being relieved. However, part-time employees accepting on a voluntary basis to work two tours of duty in any calendar day will be guaranteed a minimum of eight hours for two tours of duty, but the tours of duty can be split unevenly.
- **4.18** Part-time employees may work overtime as locally arranged in writing with due regard to Article 5.1. Overtime rates of pay will apply after eight hours in a day or 40 hours in a workweek.
- **4.19** Part-time employees will earn the hourly rate applicable to the work performed, in accordance with this Collective Agreement.
- **4.20** Part-time employees shall be included in the Health, Welfare and Pension Plans provided by the Corporation.
- **4.21** Except in emergencies, part-time employees shall not be called for duty in any seven-day period commencing Friday after they have completed 40 hours' work in such period.
- **4.22** Where work is required by the Corporation to be performed on a day which is not part of any assignment, it may be performed by an available part-time employee who would otherwise not have 40 hours of work that week.
- **4.23** Any other provision in this collective agreement relating to part-time employees shall also be applicable.
- **4.24** Part-time employees are those that have less than eight hours per day or less than 40 hours per week. The Corporation and Union agree that there shall be one category of part-time employee which is outlined in this agreement.
- **4.25** Part-time employees will be guaranteed a minimum of 20 hours per week.

- 4.26 It is not intended that part-time employees will result in the abolishment of regular assigned positions. If at any time the employment of a part-time employee does result in abolishing a regular assigned position, the Union will be given at least three months' notice and the employee affected will be entitled to maintain his basic rate of pay with a minimum of 40 hours per week, on a "Maintenance of Earnings" basis.
- **4.27** Regular part-time assignments may be established as mutually arranged, in accordance with the provisions of Articles 4.12 through 4.26, where applicable.
- **4.28** Regular part-time vacancies of 10 working days or less may be filled without the necessity of posting a notice. Vacancies of more than 10 days' duration, and new regular part-time assignments, may be filled by either posting a notice for a period of 3 days or canvassing eligible employees in seniority order at the same terminal. If there are no qualified applicants, employees will be recalled from lay-off in accordance with Article 13.
- **4.29** a) The following work rules apply to part-time employees: Articles 4.12 to 4.28, 5.3, 5.8, 8.2, 8.5 (b), 8.5 (c), 8.7, 9.23, 9.24, 9.25, 11.1, **12.1, 12.6**, 12.7 (a) 13.3 (c), 13.5, 13.13, 13.14, 13.15, **16.3, 16.4**.

Note: Any part time employee that is awarded a 12.1 temporary position will be allowed to displace onto another temporary 12.1 or 12.6. The employee will not see their status change to full time during this temporary assignment.

At the expiration of the temporary assignment, if the employee is unable to displace onto a temporary 12.1 or 12.6 position, such employee must displace under article 13.3 (c) of the Collective Agreement and will continue to be considered a part time employee.

b) Part-time employees remain governed by any articles in the Collective Agreement outlining employees' rights, benefits, and obligations other than work rules.

## ARTICLE 5 OVERTIME AND CALLS

5.1 Time worked by employees on regular assignments, continuous with, before, or after the regularly assigned hours of duty shall be considered as overtime and shall be paid at one and one-half times the hourly rate of pay in minimum increments of 15 minutes. Every effort will be made to avoid the necessity for overtime; however, when conditions necessitate, employees will perform authorized overtime work as locally arranged in writing. In the absence of a local arrangement, overtime will be offered to qualified employees in seniority order, failing which the junior qualified employee will be required to work such

overtime. An employee required to work overtime for more than two hours, continuous with completion of that employee's regular tour of eight hours' duty will be allowed without deduction of pay, 20 minutes in which to eat, immediately upon completion of two hours' overtime.

- **5.2** There shall be no overtime on overtime. Time worked in excess of 40 hours in a work week shall be paid for at time and one-half, but overtime hours paid for under Article 5.1, shall not be utilized in computing the 40 hours per week. However, up to eight hours paid for on holidays or when changing shifts may be so utilized. In addition, time paid for as arbitraries or special allowances (e.g., attending court, deadheading, travel time) shall be utilized in computing overtime when such payments apply during assigned working hours, or where such time is now included under existing articles in computations leading to overtime.
- **5.3** Time worked in excess of the regularly assigned hours, due to changing shifts, shall be paid at hourly rates, if due to application of seniority rules or where such changes in shifts are locally arranged.
- **5.4** Employees will not be required to suspend work during regular hours to absorb overtime.
- **5.5** Overtime shall be worked only by direction of proper authority. Where advanced authority is not obtainable, overtime will not be allowed unless claim is made to the proper officer within 72 hours from the time service is performed.
- **5.6** Regularly assigned employees notified, or called to work not continuous with, before, or after their regular assigned hours, shall be allowed a minimum of three hours at one and one-half times the hourly rate for three hours' work or less. This does not apply to employees whose calls are cancelled before leaving home.
- **5.7** The hourly rate for weekly rated employees is computed by dividing the weekly rate by 40.
- **5.8** Employees required to work on their assigned rest days shall be paid at one and one-half times their hourly rate with a minimum of three hours for which three hours service may be required, except:
  - (a) as otherwise provided under Article 6.
  - (b) where such work is performed by an employee moving from one assignment to another or to or from part-time status in the application of seniority or as locally arranged.
- **5.9** Overtime lists shall be updated every second week, or as otherwise locally agreed, and posted on all bulletin boards with copy to the chairperson.

## ARTICLE 6 REST DAYS

- 6.1 Employees will be assigned two rest days in each seven-day period, subject to the following:
  - (a) the work week may be staggered in accordance with the Corporation's operation requirements;
  - (b) in the instances of change in days of service, four working days' advance notice will be given to the assigned employees whose positions are affected. The Local Chairperson will be supplied with a copy of any notice.
- **6.2** The rest days shall be consecutive as far as is possible consistent with the establishment of regular relief assignments and the avoidance of working an employee on an assigned rest day. Preference shall be given to Saturday and Sunday, then to Friday and Saturday, then to Sunday and Monday. In any dispute as to the necessity of departing from the pattern of two consecutive rest days, or for granting rest days other than Friday, Saturday, Sunday, and Monday, it shall be incumbent on the Corporation to show that such departure is necessary to meet operational requirements and that otherwise additional relief service or working an employee on an assigned rest day would be involved.
- **6.3** On positions where it is not reasonably practicable to provide regular relief each week, the rest day or days for which relief is not provided may be accumulated and granted at a later date in accordance with understandings to be worked out between the designated National or Regional Representative of the Union and the proper officer of the Corporation.
- 6.4 In the event that a situation arises which makes it impracticable to assign consecutive rest days to all employees at a particular point the following procedure shall be followed:
  - (a) Full-time regular relief positions shall be established pursuant to Article 4.9;
  - (b) Possible use of rest days other than Saturday, Sunday or Monday, where these may be required under this Agreement, to be explored by the parties;
  - (c) Accumulation of rest days under Article 6.3 shall be considered;

- (d) Other suitable or practicable plans which may be suggested by either of the parties shall be considered and efforts made to come to an agreement thereon;
- (e) If the foregoing does not solve the problem, then some of the relief or extra employees may be given non-consecutive rest days;
- (f) If after all the foregoing has been done, there still remains service which can only be performed by requiring employees to work in excess of five days per week, the number of regular assignments necessary to avoid this may be made with two non-consecutive days off;
- (g) The least desirable solution of the problem would be to work some regular employees on the sixth or seventh day at overtime rates and thus withhold work from additional relief men.

## ARTICLE 7 SPARE BOARDS

**7.1** Spare Boards may be established as required under conditions to be arranged between the proper officer of the Corporation and the designated National or Regional Representative of the Union.

## ARTICLE 8 GENERAL HOLIDAYS

8.1 All employees, from their first day of work, are entitled to a holiday with pay on each of the following general holidays. When a general holiday falls on an employee's rest day, such holiday shall not be moved to the normal working day immediately following the employee's rest day.

Occasion	Atlantic	Quebec	Ontario	West	Alberta
New Year's	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Day after New Year's		$\checkmark$	✓	✓	
Family Day					✓
Good Friday	✓	$\checkmark$	✓	$\checkmark$	✓
Easter Monday	✓				
Victoria Day	✓	$\checkmark$	✓	$\checkmark$	✓
St. Jean Baptiste		$\checkmark$			
Canada Day	✓	$\checkmark$	✓	$\checkmark$	✓
Civic Holiday	✓	$\checkmark$	✓	$\checkmark$	✓
Labour Day	✓	$\checkmark$	✓	$\checkmark$	✓
National Day for	<u> </u>	<u>√</u>	<u>√</u>	✓	<u>√</u>
Truth and Reconciliation					
Thanksgiving	$\checkmark$	$\checkmark$	✓	$\checkmark$	✓
Remembrance Day	$\checkmark$		✓	$\checkmark$	$\checkmark$
Christmas	$\checkmark$	$\checkmark$	✓	$\checkmark$	✓
Boxing Day	$\checkmark$	$\checkmark$	✓	$\checkmark$	$\checkmark$

If, in any province or part thereof, a holiday is more generally recognized than any one of the holidays specified above, the signatories hereto will substitute such holiday therefore in that province or part thereof. If such signatories fail to agree that such holiday is more generally recognized, the dispute will be submitted to arbitration for final decision.

8.2

In order to qualify for any one of the holidays specified in paragraph 8.1, an employee not regularly assigned to a full-time weekly-rated position;

a) must be available for duty on such holiday if it occurs on one of his work days excluding vacation days except that this does not apply in respect of an employee who is laid off or suffering from a bona fide injury, or who is hospitalized on the holiday, or who subsequently qualifies for, weekly sickness benefits because of illness on such holiday; a regularly assigned employee who is required to work on such general holiday shall be given an advance notice of seven calendar days, except for unforeseen exigencies of the service, in which case he will be notified not later than the completion of his shift or tour of duty immediately preceding such holiday that his services will be required. The holiday pay for employees in receipt of weekly sickness benefits is deemed to be included in the weekly sickness benefits.

- 8.3 An employee regularly assigned to a full-time weekly-rated position in order to qualify for pay for any one of the holidays specified in Article 8.1;
  - (a) must have performed compensated service in the pay period in which the holiday occurs. An employee who is required to work on a general holiday shall be given an advance notice of seven calendar days except for unforeseen exigencies of the service in which case he will be notified not later than the completion of his shift or tour of duty immediately preceding the holiday that his services will be required. An employee who fails to report after having been so notified that his services will be required, will not be paid for the holiday unless his absence is due to a bona fide injury, or hospitalization, or who subsequently qualifies for weekly sickness benefits because of illness on such holiday. The holiday pay for employees in receipt of weekly sickness benefits is deemed to be included in the weekly sickness benefits.
- 8.4 A qualified employee whose vacation period coincides with any of the general holidays specified in Article 8.1 shall receive an extra day's vacation with the pay to which the employee is entitled for that general holiday.
  - (a) An assigned employee qualified under Article 8.2, or 8.3 and who is not required to work on a general holiday shall be paid eight hours' pay at the straight time rate of his regular assignment.
    - (b) Present employees on the seniority list protecting spare and relief, and extra and unassigned, and present employees who may in the future be placed on part-time, who are on the seniority list prior to January 1, 1987, qualified under Article 8.2 and who are not required to work on a general holiday shall be paid eight hours' pay at the straight time rate applicable to the position in which such employee worked his last tour of duty prior to the general holiday.
    - (c) A part-time employee having seniority on or after January 1, 1987, qualified under Article 8.2 or 8.3, and who is not required to work on a general holiday, shall be paid based on the total hours worked in the previous two pay periods divided by 20, at straight time rates, not to exceed a maximum of eight hours' pay. Straight time rate of pay will be the rate of pay of the last position worked prior to the general holiday.
- 8.6 In the application of Article 8.5 (a) for weekly-rated employees, "eight hours" pay at the straight time rate shall be deemed to be a day's pay as calculated according to Article 8.9.

8.5

- **8.7** (a) An employee who is required to work on a general holiday shall be paid, in addition to the pay provided in Article 8.5 of this Article, at a rate equal to one and one-half times his regular rate of pay for his regular shift on the holiday.
  - (b) An employee called for a specific purpose shall be guaranteed a minimum of three hours for which three hours of service may be required at a rate equal to one and one-half times his regular rate of pay, but he shall not be required to perform routine work to make up such minimum time.
- **8.8** Shifts or tours of duty commencing between 0001 hours on the general holiday and 2359 hours on the night of the general holiday, both times inclusive, shall be considered as work on that holiday.
- **8.9** The daily rate of pay for weekly-rated employees shall be the weekly rate divided by five.

## ARTICLE 9 VACATIONS

- **9.1** An employee who, at the beginning of the calendar year, is not qualified for vacation under Article 9.2 hereof, shall be allowed one working days' vacation with pay for each 25 days **worked**, or major portion thereof, during the preceding calendar year, with a maximum of 10 working days until qualifying for further vacation under Article 9.2.
- **9.2** Subject to the provisions of Note 1 below, an employee who, at the beginning of the calendar year, has maintained a continuous employment relationship for at least three years, shall have his vacation scheduled on the basis of one working days' vacation with pay for each 16-2/3 days **worked**, or major portion thereof, during the preceding calendar year, with a maximum of 15 working days; in subsequent years, he will continue vacation entitlement on the foregoing basis until qualifying for additional vacation under Article 9.3.
  - **NOTE 1:** An employee covered by Article 9.2 will be entitled to vacation on the basis outlined herein on his fourth or subsequent service anniversary date otherwise his vacation entitlement will be calculated as set out in Article 9.1. Any vacation granted for which the employee does not subsequently qualify will be deducted from the employee's vacation entitlement in the next calendar year. If such employee leaves the service for any reason prior to his next vacation, the adjustment will be made at time of leaving.

- **9.3** Subject to the provisions of Note 2 below, an employee who, at the beginning of the calendar year, has maintained a continuous employment relationship for at least 9 years, shall have his vacation scheduled on the basis of one working days' vacation with pay for each 12 1/2 days **worked**, or major portion thereof, during the preceding calendar year, with a maximum of 20 working days; in subsequent years, he will continue vacation entitlement on the foregoing basis until qualifying for additional vacation under Article9.4.
  - **NOTE 2** An employee covered by Article 9.3 will be entitled to vacation on the basis outlined therein if on his tenth or subsequent service anniversary date otherwise, his vacation entitlement will be calculated as set out in Article 9.2. Any vacation granted for which the employee does not subsequently qualify will be deducted from the employee's vacation entitlement in the next calendar year. If such employee leaves the service for any reason prior to his next vacation, the adjustment will be made at time of leaving.
- **9.4** Subject to the provisions of Note 3 below, an employee who, at the beginning of the calendar year, has maintained a continuous employment relationship for at least 19 years shall have his vacation scheduled on the basis of one working days' vacation with pay for each 10 days **worked**, or major portion thereof, during the preceding calendar year, with a maximum of 25 working days; in subsequent years, he will continue vacation entitlement on the foregoing basis until qualifying for additional vacation under Article 9.5.
  - **NOTE 3**: An employee covered by Article 9.4 will be entitled to vacation on the basis outlined therein if on his twentieth or subsequent service anniversary date; otherwise, his vacation entitlement will be calculated as set out in Article 9.3. Any vacation granted for which the employee does not subsequently qualify will be deducted from the employee's vacation entitlement in the next calendar year. If such employee leaves the service for any reason prior to his next vacation, the adjustment will be made at time of leaving.
- **9.5** Subject to the provisions of Note 4 below, an employee who, at the beginning of the calendar year, has maintained a continuous employment relationship for at least 26 years shall have his vacation scheduled on the basis of one working days' vacation with pay for each 8 1/3 days **worked**, or major portion thereof, during the preceding calendar year, with a maximum of 30 working days.

- **NOTE 4**: An employee covered by Article 9.5 will be entitled to vacation on the basis outlined therein if on his twenty-seventh or subsequent service anniversary date otherwise, his vacation entitlement will be calculated as set out in Article 9.4. Any vacation granted for which the employee does not subsequently qualify will be deducted from the employee's vacation entitlement in the next calendar year. If such employee leaves the service for any reason prior to his next vacation, the adjustment will be made at time of leaving.
- **9.6** An employee who while on annual vacation, becomes ill or is injured shall have the right to terminate (temporarily) his vacation and be placed on short term disability. An employee who is again fit for duty shall immediately so inform the corporate officer in charge and will continue his vacation if within his scheduled dates. If the remaining vacation falls outside the employee's scheduled dates, such vacation will be rescheduled as may be locally arranged.
- **9.7** An employee who, due to sickness or injury, or on an authorized leave of absence for union business, is unable to take or complete his annual vacation in that year shall, at the option of that employee, have the right to have such vacation carried to the following year.
- **9.8** An employee who is entitled to vacation shall take same at the time scheduled. If, however, it becomes necessary for the Corporation to reschedule an employee's scheduled vacation dates, he shall be given at least 15 working days' advance notice of such rescheduling and will be paid at the rate of time and one half his regular rate of wages for all work performed during the scheduled vacation period. The rescheduled vacation with pay to which he is entitled will be granted at a locally agreed upon later date. This Article 9.10 does not apply where rescheduling is a result of an employee exercising his seniority to a position covered by another vacation schedule.
- **9.9** (a) Provided an employee render compensated working service in any calendar year, time off duty account bona fide illness, injury, union business in accordance with Articles 17.2, 17.3 and 17.4, called to court as a witness, or for uncompensated jury duty, not exceeding a total of 150 days in any calendar year, shall be included in the computation of service for vacation purpose.

(b) An employee who takes an authorized leave of absence for maternity and/or parental in accordance with the terms and conditions of the Canada Labour Code, Part III shall have such period included in the computation of service for vacation purposes.

- **9.10** Vacation days shall be exclusive of the assigned rest days and the legal holidays and bereavement days specified in Articles 6, 8 and 29 respectively.
- **9.11** Days worked on any position covered by a similar Vacation Agreement will be counted as service for vacation purposes under this Agreement.
- **9.12** An employee will be compensated for vacation at the rate of the position which he would have been filling during such vacation period.
- **9.13** An employee terminating his employment for any reason at a time when an unused period of vacation with pay stands to his credit shall be allowed vacation calculated to the date of his leaving the service, as provided for in Articles 9.1, 9.2, 9.3, 9.4, and 9.5, and, if not granted, will be allowed pay in lieu thereof.
- 9.14 (a) In the event an employee is laid off, they may elect either;
  - (i) To receive payment in lieu of vacation for all outstanding vacation due to them at the beginning of the current calendar year, or;
  - (ii) To take the vacation at the time allotted to them pursuant to Article 9.20 or 9.21.
  - (b) If the employee is laid off at the time the vacation was scheduled, they shall receive payment in lieu of vacation for all outstanding vacation due to them.
  - (c) In the event there is a payment made in lieu of vacation, the number of days of vacation represented by the payment will be included in the computation of cumulative compensated service.
- **9.15** An employee who (1) leaves the service of his own accord, (2) is dismissed for cause and not reinstated in his former seniority standing within four years of date of such dismissal, will, if subsequently returned to the service be required to again qualify for vacation with pay as provided in Articles 9.1, 9.2, 9.3, 9.4, and 9.5.
- **9.16** An employee who has become entitled to a vacation with pay shall be granted such vacation within a 12-month period immediately following the completion of a calendar year of employment in respect of which the employee became entitled to the vacation.
- **9.17** Applications for annual vacations from employees shall be filed prior to February 1st in the group where the employees are currently holding a position.

- **9.18** Applications filed prior to February 1st, insofar as it is practicable to do so, will be allotted vacation during the summer season, in order of seniority of applicants, and unless otherwise locally arranged, the vacation period shall be continuous. Applicants will be advised in writing in February of dates allotted them in the group where applicants are currently holding a position, and unless otherwise locally arranged, employees must take their vacation at the time allotted.
- **9.19** Unless otherwise locally arranged, employees who do not apply for vacation prior to February 1st, shall be required to take their vacation at a time to be prescribed by the Corporation.
- **9.20** Notwithstanding Article 12.1 and 12.6, the Manager responsible for vacation allocation and the Local representative or his/her delegate will meet in

November each year to review vacation scheduling in order to, as far as practicable, make local arrangements to carry on work while members are on vacation without incurring additional expense for the Corporation.

Should such arrangements result in the establishment of a vacation relief position, it shall be bulletined in accordance with Article 12. If this is not practicable (first sentence), employees engaged temporarily, or employees temporarily promoted from one position to another, to provide vacation relief, will, if definitely assigned to fulfil the duties and responsibilities of a higher rated position, be paid the schedule rate applicable to such position.

- **9.21** Employees who exercise their seniority after vacation dates are allotted and transfer from the group to which assigned when vacation dates were allotted will be required to return to such group at least 30 days prior to the scheduled start date of the vacation, failing which they must take their vacation at a time as locally arranged.
- **9.22** Part-time employees will be compensated for vacation on the basis of a percentage of their previous year's gross earnings, the percentage amounts to be determined based on entitlement as specified under Articles 9.1 through 9.5 of this agreement, that is, 2% per week of vacation:

2 weeks' vacation	-	4%
3 weeks' vacation	-	6%
4 weeks' vacation	-	8%
5 weeks' vacation	-	10%
6 weeks' vacation	-	12%

**9.23** Employees promoted to official or excepted positions after vacation dates are allotted and transfer from the group to which assigned when vacation dates were allotted will be required to return to such group at least 30 days prior to the scheduled start date of the vacation, failing which they must take their vacation at a time as locally arranged.

## ARTICLE 10 SENIORITY GROUPINGS

**10.1** For the purpose of seniority, employees will be grouped as:

VIA Atlantic VIA Ontario VIA Quebec VIA West

## ARTICLE 11 SENIORITY

- 11.1 An employee will be considered on probation until they have completed eighty (80) days of actual work in the service of the Corporation. The probationary period commences following the successful completion of training. The employees may be removed for cause, which, in the opinion of the Corporation, renders the employee unsuitable for its service during such period. A Union representative or designate will be invited to participate in the meeting which may be in person or by phone when an employee on probation is discharged or disciplined. The probationary employee shall have access to the grievance procedure.
- **11.2** A seniority list will be maintained for each seniority group showing seniority numbers, names, positions, location, and date of last entry into the Corporation's service in a position covered by such seniority group, from which date seniority will accumulate. Seniority lists will be maintained by the Corporation and a copy furnished to the designated National or Regional Representative of the Union and the Local Chairperson concerned in February, June, and October of each year.

A copy of the appropriate seniority list will be posted each February in a location suitable for the employees concerned. The date the seniority list is posted at each location will be shown on the seniority list.

11.3 (a) Employees service dates will be established upon initially entering training. The name of an employee shall be placed on the seniority list immediately upon successful completion of training and will be retroactive to their service date. Due to the impact of varying start times to work shifts/schedules for establishing seniority, the following process will be used to determine seniority when more than 1 person commences work in the same seniority group:

With the designated union representative for the group present during the prescribed training period as outlined in the collective agreement, a random draw with all the names of current entry level participants will be conducted by the union representative in order to establish seniority.

Employees will remain on the seniority list providing they have compensated service under Agreement No. 1 or No. 2 within 24 months from the day of last entry into service on a position covered by Agreement No. 1 or No. 2, and any period of 24 months thereafter, otherwise they will forfeit their seniority, their name will be removed from the seniority list and their employment will be terminated. The above period of 24 months will commence after the weekly lay-off benefits terminate.

If the affected employee wishes to maintain his seniority, they must so notify

the Department Director, or his designate, in writing prior to the completion of the 24-month period. This period will be extended for a further 12 months. If no actual work is performed by the employee for the Corporation within the further 12-month period, the employee then will forfeit his seniority, his name will be removed from the seniority list and his employment with the Corporation will be terminated.

Furthermore, when an employee is on an authorized leave of absence for any reason (i.e., illness, injury, Union leave, WCB and others), he is considered to be as having been on compensated service. The designated National or Regional Representative of the Union shall be provided with a list of the names of those employees who have been removed from the seniority list through the application of this article.

(b) Employees while filling positions under this Agreement who accept positions covered by another wage agreement may continue to fill such positions for a continuous period up to 6 months without loss of seniority. Provided they can hold work on their own seniority region, such employees shall exercise their seniority at, or prior to, the expiration of such 6-month period, or forfeit their seniority rights under this Agreement.

An employee who chooses to remain within the new collective agreement will be eligible for Job Security under the Supplemental Agreement after (2) two years or more of service in the new collective agreement. As a result, Weekly Lay-off Benefits will be based on the employee's service in the new agreement.

Upon reaching two years' service in the new agreement, transferred employees will be entitled to Weekly Lay-off Benefits based on the calculation of service from the date of entry into the Corporation's service as an employee.

**NOTE:** The present article will not apply to employees who are transferred from one Agreement to another under a duty to accommodate agreement. Accommodated employees' service will be calculated as though they had not transferred agreements and in accordance with the Supplemental Agreement.

Employees currently on Job Security benefits will remain eligible for such benefits in accordance with the Supplemental Agreement.

- **11.4** Protests respecting seniority status must be submitted in writing within 60 calendar days from the date seniority lists are posted. When proof of error is presented by an employee or his representative, such error will be corrected and when so corrected the agreed upon seniority date shall be final. No change shall be made in the existing seniority status of an employee unless concurred by the designated National or Regional Representative of the Union.
- 11.5 No change shall be made in the seniority date accredited an employee which has appeared on two consecutive annual seniority lists unless the seniority date appearing on such lists was protested in writing within the 60-calendar day period allowed for correctional purposes. Names which have not appeared on two consecutive annual seniority lists shall not be restored to such seniority lists except in accordance with Article 11.13 or by agreement with the designated National or Regional Representative of the Union. A supplemental bulletin will be issued by the Corporation and posted by June 30<sup>th</sup> of each year showing any corrections to the seniority list as provided for above.
- 11.6 An employee with less than one year's seniority who, while filling a position under this Agreement, accepts a non-supervisory position under another wage agreement shall forfeit his seniority under this Agreement and his name shall be removed from the seniority list. This shall not apply when the employee accepts temporary and/or relief work under another wage agreement but should such temporary and/or relief work extend into a continuous period exceeding six months, he shall forfeit his seniority under this Agreement and his name shall be removed from the seniority.

- 11.7 An employee with one year's seniority or more who, while filling a position under this Agreement, accepts non-supervisory work under another wage agreement shall be permitted to perform such work for a continuous period up to six months without loss of seniority. However, provided he can hold work in his own seniority group, he must return to such group at or prior to the expiration of such six-month period or forfeit his seniority list. After return from work under another wage agreement, the employee must remain on a position covered by this Agreement for a continuous period of at least six months. If he returns to work under another wage agreement before the expiration of such six months, except when required for emergency work under another wage agreement, he will forfeit his seniority under this Agreement.
- **11.8** The provisions of Articles 11.6 and 11.7 shall not apply to an employee who, while holding seniority rights under another wage agreement, obtains employment and establishes seniority rights under this Agreement. If such an employee, while filling a position under this Agreement, exercises his seniority under the provisions of another wage agreement, his name shall be removed from the seniority list. An employee shall not be regarded as having exercised seniority rights when used for emergency service only.

## 11.9 (a) Permanent Positions

An employee holding seniority under this Agreement and who is presently filling or who may in the future be promoted to an official permanent position with the Corporation which is excepted from any provision of this or any Collective Agreement, will have his or her name continued on the seniority list of the group from which promoted at his or her home seniority terminal and will retain seniority rights and continue to accumulate seniority while so employed for the duration of six (6) cumulative months. The employee shall be required to pay unions dues at the rate of the last position held prior to being staffed out for the duration of six (6) cumulative months for the sole purpose of protecting seniority. The corporation will be required to provide written notice to the Regional Representative of promotions at the time when the employee is staffed out.

Thereafter, such employee will cease to accumulate any further seniority until such employee returns to a position within the bargaining unit. The period of six cumulative months may be extended through a managementunion accord. When an employee who has not forfeited his seniority under the above provisions, is released from such excepted employment, except at his own request or as provided in Article 12.19, he may exercise his seniority rights to any position in his seniority group which he is qualified to fill. He must make his choice of a position, in writing, within 10 calendar days from the date of release from excepted employment and commence work on such position within 30 calendar days from the date of release from excepted employment. Failing this, he shall forfeit his seniority and his name shall be removed from the seniority list.

- **<u>NOTE:</u>** When an employee is promoted to an excepted position his position will be filled in accordance with Article 12.1. When released from the excepted position he must exercise seniority within his seniority group for which he is qualified to fill.
- **NOTE:** If the employee who is temporarily promoted does not advise the Regional Representative in writing before the end of the next pay period of the date, he is occupying a management position, the employee's seniority will be forfeited as of the date he started on the excepted management position.

## (b) Temporary Positions

An employee holding seniority under this Agreement and who is presently filling or who may in the future be promoted temporarily to an official or any Temporary position with the Corporation which is excepted from any provision of this or any Collective Agreement, will have his name continued on the seniority list of the group from which promoted at his home seniority terminal and will retain seniority rights and continue to accumulate seniority while so employed for a period of twelve (12) consecutive months. The employee shall be required to pay unions dues at the rate of the last position held prior to being staffed out for the duration of the twelve (12) consecutive months for the sole purpose of protecting seniority. Thereafter, such employee will cease to accumulate any further seniority until he returns to a position within the bargaining unit. The appropriate officer of the Corporation shall advise the Regional Representative in writing of such temporary promotions and the duration thereof at the time when the employee is staffed out. The period of twelve consecutive months may be extended through a mutual agreement

When an employee who has not forfeited his seniority under the above provisions, is released from such excepted employment, except at his own request or as provided in Article 12.19, he may exercise his seniority rights to any position in his seniority group which he is qualified to fill. He must make his choice of a position, in writing, within 10 calendar days from the date of release from excepted employment and commence work on such position within 30 calendar days from the date of release from excepted employment. Failing this, he shall forfeit his seniority and his name shall be removed from the seniority list.

- **NOTE:** When an employee is temporarily promoted to an excepted position for less than 90 days his position will be filled in accordance with Article 12.6. When released from the excepted position he must return to his regular assignment.
- **NOTE:** In the application of this paragraph, should an employee holding a non-scheduled, official, or excepted position be set back to a position covered by this Collective Agreement for a period of less than 90 days such time will be considered as part of the twelve (12) consecutive months.
- **NOTE:** If the employee who is promoted temporarily does not advise the Regional Representative in writing before the end of the next pay period of the date, he is occupying a management position, the employee's seniority will be forfeited as of the date he started on the excepted management position.

If the 12-month temporary position becomes permanent, the seniority will freeze at the end of the 6-month period.

## (c) Temporary Training Instructor

An employee holding seniority under this agreement and who is presently filling or who may in the future be promoted to an official or excepted position as temporary training instructor with the Corporation which is excepted from any provision or any Collective Agreement will have his or her name continued on the seniority list of the group for a period of twelve months from which promoted at his or her home seniority group and will retain seniority rights and continue to accumulate seniority while so employed. The employee shall be required to pay unions dues at the rate of the last position held prior to being staffed out for the duration of the twelve (12) consecutive months for the sole purpose of protecting seniority. The Corporation will notify the Regional representative in writing of such promotions at the time the employee is staffed out. When an employee who has not forfeited his seniority under the above provisions, is released from such excepted employment, except at his own request or as provided in Article 12.19, he may exercise his seniority rights to any position in his seniority group which he is qualified to fill. He must make his choice of a position, in writing, within 10 calendar days from the date of release from excepted employment and commence work on such position within 30 calendar days from the date of release from excepted employment. Failing this, he shall forfeit his seniority and his name shall be removed from the seniority list.

- **NOTE:** When an employee is temporarily promoted to an excepted position for less than 90 days his position will be filled in accordance with Article 12.6. When released from the excepted position he must return to his regular assignment.
- **<u>NOTE</u>**: If the employee who is promoted temporarily does not advise the Regional Representative in writing before the end of the next pay period of the date, he is occupying a management position, the employee's seniority will be forfeited.

If the 12-month temporary position becomes permanent, the seniority will freeze at the end of the 6-month period.

- **11.10** The name of an employee who has been or is transferred from a position covered by this Agreement to a position with an association affiliated with the Corporation, e.g. The Railway Association of Canada, will be continued on the seniority list for the group in which he holds seniority and shall continue to accumulate seniority while so employed. Such individual, when released from such employment, except as provided in Article 12.19, may exercise his seniority rights to any position in his seniority group which he is qualified to fill. He must make his choice of a position, in writing, within 10 calendar days from the date of release from such other employment and commence work on such position within 30 calendar days from the date of release from the seniority list.
- **11.11** The name of an employee transferred with his work from a staff covered by this Agreement to a staff not covered by this Agreement, shall be removed from the seniority list.

- **11.12** The seniority status of an employee transferred with his work from a staff not covered by this Agreement to a staff covered by this Agreement shall be decided by agreement between the proper officer of the Corporation and the designated National or Regional Representative of the Union. The basis of such decision shall be the seniority to which he would have been entitled had his service on such other staff been governed by the terms of this Agreement.
- 11.13 An employee who has been discharged and is subsequently returned to the service on a position covered by this Agreement will only be allowed seniority from the date of his return to the service, unless reinstated with his former seniority status. An employee who is not reinstated with his former seniority status within two years of the date of his discharge may only be so reinstated by agreement between the proper officer of the Corporation and the designated National or Regional Representative of the Union.

## ARTICLE 12 BULLETINING AND FILLING OF POSITIONS

- **12.1** Temporary vacancies, newly created positions, or seasonal positions, any of which are known to be of more than 90 calendar days' duration, and vacancies in permanent positions will be bulletined on the Region to the seniority group concerned.
- **12.2** When required, Regional bulletins will be issued every third Wednesday. Bulletins will be posted promptly for a period of 12 calendar days in places accessible to all employees affected and a copy of each bulletin will be furnished to the Local Chairperson concerned.
- **12.3** All bulletins will show classification and location of the position, general description of duties, necessary qualifications (where applicable), rate of pay, level of the position, hours of assignment including meal period, assigned rest days, the approximate date of commencement for seasonal and temporary assignments and their approximate duration. Where the nature of the work will require the successful applicant to perform his duties outside, such information will be specified in the bulletin.

Employees other than those referred to in Article 11.9 desiring such position will submit written application showing seniority number, present classification and location, together with their qualifications. Except as provided in Article 12.4, applications must be filed to reach the designated officer not later than the twelfth day after the date of bulletin. As evidence that an application has been submitted each applicant must forward a copy of his application to his Local Chairperson.

12.4 A permanent position shall be declared vacant and bulletined only to the seniority group at the station or terminal affect, when the regularly assigned starting time or spread of hours is changed two hours but less than eight hours or assigned rest day or days are changed. Such position shall be awarded to the qualified senior employee at such station or terminal affected who makes written application therefore within five calendar days from the date the bulletin is posted, and subsequent vacancies will be advertised in the same manner.

An employee displaced as a result of the foregoing must within three calendar days of being displaced exercise his seniority rights to another position which he is qualified to fill in his own seniority group at his station or terminal. When the starting time or spread of hours of a position is changed eight hours or more, the position will be bulletined to the region.

- **12.5** When the assigned starting time or spread of hours of a position is changed one hour or more but less than two hours, the meal period is changed, the incumbent of such position may exercise his seniority to another position within the same classification at the same office or station. The employees affected thereby will exercise their seniority within the same classification at the same office or station at the same office or station. Such employees will not be considered as displaced within the meaning of Article 13.
- **12.6** Temporary vacancies, newly created positions, and seasonal positions, when known to be for 90-calendar days' duration or less, will not be bulletined. However, suitable advice notice will be posted, as required, at the station or terminal affected and a copy of the advice notice will be sent to the Local Chairperson. Such position shall be awarded to the qualified senior employee on the Region who makes application therefore within five calendar days from the date notice is posted. The successful applicant shall be permitted to assume the temporary vacancy within 10 days from the date the advice notice is posted. Applications from regularly assigned employees will only be accepted when it is known the vacancy is for more than 10 working days and when it involves an increase in rate of pay, or a change in shift, or rest day or days. When other qualified employees are

available regularly assigned employees will not be allowed to commence work on a temporary vacancy and work their regular assignment on the same day.

Note: For a ninety (90) day period expiring one week following labour day, employees awarded a 12.6 assignment, will be required to remain on the assignment for the length indicated. Exception will be made if the employee bids on and obtains a vacant permanent position posted under article 12.1 or on a training posted under article 16.3, 16.4 or any other 12.6 vacancy within the department or is displaced.

- **12.7** Temporary vacancies of 10 working days or less, and vacancies in other positions pending occupancy by the successful applicant may be filled without the necessity of advice notice or bulletin:
  - (a) first by a qualified part-time employee who has not completed 40 hours of work for any particular week;
  - (b) then by a senior qualified regularly assigned employee at the same station or terminal who desires such work.

An employee filling a temporary vacancy pending occupancy by the successful applicant will not be subject to displacement during the first 30 days of occupancy. When it is known that a temporary vacancy will occur, employees desiring the position may be required, as locally arranged, to make their intentions known some time prior to the starting time of the vacancy. The employee, so assigned, will not be subject to displacement during such period, except by a senior qualified employee unable to hold work at the station or terminal affected.

**12.8** An employee, who has applied for a bulletined position, may cancel his application provided written cancellation reaches the designated officer not later than the twelfth calendar day after date of a Regional bulletin or the fifth calendar day in the case of a station bulletin. As evidence that the application has been cancelled, the employee must forward a copy of his cancellation to the Local Chairperson.

Unless there is no other qualified applicant, an employee vacating a position will not be considered for such position until it again becomes vacant.

**12.9** Where no applications are received from qualified employees in the seniority group in which a vacancy occurs, and no qualified employees are available on the Region laid-off list, a written application from the qualified senior employee from another seniority group will be given preference. Such an employee will accumulate seniority rights in his new group from the date he starts work on a position in that new seniority group. He will also retain all rights in his former group until such time he exercises his seniority in the new seniority group. Upon returning to his former seniority group, he will forfeit his rights in the group to which he had transferred.

- **12.10** An employee who, in accordance with Article 12.9 transfers from one group to another and later transfers to a third group will forfeit his seniority in the original group. Similarly, an employee transferred from the second seniority group to an excepted position will forfeit his seniority in the first seniority group.
- 12.11 In the event that there is an unfilled vacancy for which there is no qualified applicant, a qualified employee at the station or terminal may be required to fill such a position. The qualified employee will be selected in reverse seniority order on a rotating basis to distribute the required work. In such cases, the Corporation will commence the training of another employee for the position immediately so that the employee required to fill the position may be returned to his regular assignment as soon as is practicable and shall be able to resume his former position after 45 calendar days. The Corporation shall inform the Local Chairperson under whose jurisdiction the employee comes that this Article has been invoked.
- **12.12** When a vacancy or a new position is to be filled, it shall be awarded to the senior applicant who has the qualifications required to perform the work. Management will be the judge of qualifications subject to the right of appeal by the employee and/or the Union. The name of the appointee and his seniority will be shown on the next bulletin. Should there be no qualified applicant, management may consider the applications, in seniority order.
- **12.13** Employees shall be permitted to assume positions bulletined in accordance with Article 12.1 to which appointed within 21 calendar days of the date of bulletin making the appointment and must assume such position within 45 calendar days of such appointment or on completion of their present temporary assignment.
- **12.14** A regularly assigned employee who is assigned to a temporary vacancy of any duration or to any position of 90-calendar days' duration or less may, upon completion of such temporary assignment, displace junior employee on any other temporary assignment, except as provided for in Article 12.7 before returning to his regularly assigned position.

- **12.15** An employee, returning from vacation or leave of absence (except as provided in Article 11.10), shall immediately resume his former position and within five working days of his return may exercise his seniority to any position bulletined in accordance with Articles 12.1, 12.4 or 12.6 during his absence. When displacing, in accordance with Article 12.6, employees will only be permitted to displace at their station or terminal. Employees thereby displaced will return to their former assignments or may exercise their seniority rights to any position awarded under Articles 12.1, 12.4 and 12.6 to a junior employee during the period between their appointment and subsequent displacement.
- 12.16 An employee, who is assigned to a position by bulletin, will receive a full explanation of the duties of the position and must demonstrate his ability to perform the work within a reasonable probationary period up to 30 working days, the length of time dependent upon the character of the work. Any extension of time beyond 30 working days shall be locally arranged.

Failing to demonstrate his ability to do the work, he shall be returned to his former position without loss of seniority and the employee so displaced will be allowed to exercise his seniority. When an employee who has been assigned to a position by bulletin fails to demonstrate his ability to perform the work, the position will be re-bulletined.

- **12.17** When a senior applicant is not awarded a bulletined position, he may appeal the appointment, in writing, within 21 calendar days of such appointment through the grievance procedure. After making an appeal, he may be required or shall at the request of the Local Chairperson or authorized Committee person be allowed to demonstrate his qualifications for the position. The Local Chairperson or authorized Committee person may be present at such demonstration.
- **12.18** An employee, removed from a position to which he had been appointed, as a result of a grievance filed by a senior employee, may return to his former position, or exercise his seniority rights to any position for which he is qualified, awarded to a junior employee during the period between his appointment and subsequent removal and the employee so displaced will be allowed to exercise his seniority.
- **12.19** Unless otherwise locally arranged, an employee who is removed from his regular position as a disciplinary measure will not be permitted to displace any regularly assigned employee within the first 12 months of cumulative compensated service after the discipline was assessed but will be permitted to apply for any vacancies within that group.

Upon completion of the said 12-month period, the employee may apply for his former position if it is vacant at the time. If his former position is not vacant, he may displace a junior employee in the same position at the end of the 23 months, but not later than 24 months after the assessment of discipline.

#### ARTICLE 13 STAFF REDUCTION, DISPLACEMENT AND RECALL TO SERVICE

- **13.1** When staffs are reduced, senior employees with sufficient ability to perform the work will be retained.
- **13.2** In instances of staff reduction, 14 calendar days' advance notice will be given to regularly assigned employees whose positions are to be abolished, except in the event of a strike or work stoppage by employees in the railway industry, in which case a shorter notice may be given. The Local Chairperson will be supplied with a copy of any notice.
- **13.3** An employee whose position is abolished or who is displaced from his permanent position shall:
  - (a) Displace a junior employee in his own seniority group on a permanent position, for whose position he is qualified.
  - (b) Displace a junior employee in his own seniority group on a temporary position, for whose position he is qualified if unable to hold a permanent position after exhausting seniority rights.

Employees who are displaced from their permanent position pursuant to Article 13.3 of the Collective Agreement and who are not qualified to hold another position, in or out of the skill set, with a rate of pay equal to or greater than their previous rate of pay will be entitled to the following opportunity :

- i) One period of familiarization for a position within their skill set to which their seniority would entitle them;
- ii) The familiarization period shall not exceed 5 working days and will involve informal on the job training;
- iii) Time spent on training will be considered as time worked.

#### **NOTE:** The skill sets are identified at Appendix R.

or

(c) After exhausting his seniority rights at his home station or terminal, he may elect to protect part-time work in his own seniority group at his present station or terminal or at any station or terminal on his region at which he has previously been laid off or displaced providing work is available at such point. (See Article 4.13 note).

Such an employee shall forfeit his seniority, if he does not notify the officer in charge and the Local Chairperson in writing of his choice within seven (7) calendar days from the date of displacement or abolition of his position.

An employee who does not elect (c) above and has exhausted his seniority rights under his own job security eligibility territory, will have his name placed on his regional laid-off list.

13.4 An employee, who has signified his intention to displace a junior employee, shall forfeit his seniority and his name shall be removed from the seniority list and his employment will be terminated if he fails or refuses to commence work at the next effective date on the regularly assigned position, he has chosen within seven (7) calendar days of making his choice, or within three (3) calendar days of exercising his seniority to a temporary assignment.

An employee holding a permanent position who is completing or being displaced from a temporary position may displace a junior employee on another temporary position for whose position he is qualified. If he does not hold a permanent position he must displace to a permanent position if he is able to do so.

- **13.5** An employee, who has signified his intention to remain available for parttime work shall be governed by Articles 4.12 to 4.27. The foregoing provisions shall also apply to an employee hired for part-timework.
- **13.6** A senior employee allowed to displace a junior employee shall receive a full explanation of the duties of the position and must demonstrate his ability to perform the work within a reasonable probationary period up to 30 working days, the length of time dependent upon the character of the work. Any extension of time beyond 30 working days shall be locally arranged. The provisions of Article 12.17 may be applied in cases when an employee is not allowed to displace.

- **13.7** An employee who has exercised his seniority in accordance with this Article and fails to show necessary qualifications for the position he has chosen, will be required to vacate such position. He may again, displace a junior employee for whose position it is considered he is qualified. The employee he originally displaced, and other employees displaced as a direct consequence thereof, shall return to their former positions.
- **13.8** When an employee is on leave of absence or vacation at the time his position is abolished or he is displaced, the time limits specified in this Article will apply from the time he reports for duty.
- **13.9** An employee, who fails to comply with Articles 13.3, 13.4 and/or 13.5 because of illness, or other cause for which leave of absence has been granted, shall not lose his seniority.
- **13.10** A laid-off employee must register his name and address, in writing at time of layoff, with his immediate supervisory officer and his Local Chairperson. He must also advise in writing, the proper officer of the Corporation and the Local Chairperson of any change of address. An employee who fails to comply with either of these requirements shall forfeit his seniority and his name shall be removed from the seniority list.
- **13.11** When a vacancy is not filled in accordance with Article 12.9, a laid-off employee, if qualified, shall be given preference of employment in seniority order in filling new positions or vacancies in other than his own seniority group.
- **13.12** A laid-off employee, who accepts works in a seniority group other than his own, will accumulate seniority from the date he commences work in such group. He will retain full seniority rights in his former group until such time as he refuses to accept a recall to such former group until such time as he refuses to accept a recall to such former group. Upon returning to his former group, he will forfeit all rights in the group to which he had transferred.
- **13.13** A laid-off employee shall, if qualified, be recalled to service in order of seniority when a vacancy in his seniority group remains unfilled after having been bulletined. An employee, recalled from layoff, shall be notified by registered mail to the last address on record with the Corporation.
- **13.14** A laid-off employee, subject to recall, will not be required to report for duty providing that:
  - (a) It is definitely known that the duration of the work will not exceed 90 calendar days and another junior qualified laid-off employee is available, or

- (b) The position available is not in his own job security eligibility territory.
- **<u>NOTE:</u>** This Article does not constitute a guarantee of 90 days of employment.

In either event, the employee concerned must give written advice of his intentions to his immediate supervisor immediately upon receipt of notification to resume duty.

Employees who elect to protect part-time work in accordance with Article 13.3(b), and who are subsequently laid off, must accept to return from layoff to part-time work on a senior may, junior must basis. Junior employees who refuse to return to work when called will not be offered any other part-time work during their period of lay-off.

Employees who are hired for part-time work will be required to accept parttime work when called on a senior may, junior must basis.

**13.15** A laid-off employee who fails to report for duty or to give a satisfactory reason to the Corporation, in writing, for not doing so within 10 calendar days from date of notification, shall forfeit his seniority, his name shall be removed from the seniority list and his employment will be terminated.

# ARTICLE 14 TRANSFERS

**14.1** When work is transferred from one seniority group to another, employees shall be transferred with it as may be mutually agreed between the proper officer of the Corporation and the designated National or Regional Representative of the Union. The names of such employees shall be removed from the seniority list of the group from which transferred and included with full seniority on the list of the group to which transferred.

## ARTICLE 15 REHABILITATION

- **15.1** When mutually agreed between the Director, Human Resources or designate, and the designated National or Regional Representative of the Union, an employee who has become unfit\* to follow his usual occupation may
  - (a) displace a junior employee in his own seniority group for whose position he is qualified, or
  - (b) be placed, when mutually agreed between the proper officer of the

Corporation and the designated National or Regional Representative of the Union, in a position on his Region, notwithstanding that it may be necessary to displace an able-bodied employee to provide suitable employment for him. Such mutual agreement will not be unreasonably or arbitrarily withheld.

- (c) if, after the application of sub-paragraphs (a) and (b) above, employees are still not able to hold work, they may be trained, providing they have the suitability and adaptability, to fill a vacant assignment or to displace a junior employee.
- \*NOTE: The recognized Medical Department will determine an employee's fitness to follow his usual occupation. The designated National or Regional Representative of the Union will be advised

In the event of a dispute regarding the Corporation medical department's determination, the employee will be entitled to one independent medical assessment, selected, and paid for by the Corporation in line with the Corporation's current medical department's policy. Such assessment becomes binding for both parties.

- **15.2** In dealing with incapacitated employees, seniority shall govern in respect of preference of shift and employment.
- **15.3** A rehabilitated employee placed on a position shall not be displaced by an able-bodied employee so long as he remains on such position, except when a senior employee is otherwise unable to hold a position in his seniority group. Should he subsequently recuperate he shall be subject to displacement in which case he shall exercise his seniority rights.

# ARTICLE 16 TRAINING

- **16.1** Newly hired employees must remain on the position(s) within their department they were hired for at least 6 months of cumulative compensated service before being able to apply for a training or on another position.
- **16.2** Shall be encouraged to learn the duties of other positions and every opportunity shall be afforded them to learn the work of such positions in their own time, and during regular working hours when it will not unduly interfere with the performance of their regularly assigned duties.

The supervisory officer may for this purpose arrange with the interested

employees to exchange positions for temporary periods without affecting the rates of pay of the employees concerned. The Local Chairperson of the Union will be informed when employees exchange positions in accordance with this Article.

#### 16.3 Training During Normal Working Hours

An employee required by the Corporation to take training during his normal working hours will be paid his regular rate of pay while in training.

## **Training Outside Normal Working Hours**

An employee required by the Corporation to take training outside his normal working hours will be compensated at his regular rate of pay while in training, except that on any day when the Corporation requires an employee to take training in addition to working his regular assignment, he shall be compensated for all such combined time, in excess of eight hours, at punitive rates.

# **Voluntary Training**

Where training facilities are provided by the Corporation on a voluntary basis an employee taking advantage of such training will not be compensated.

16.4 The Corporation shall have training courses which shall be sufficient to allow the employees opportunities to upgrade their knowledge and skills. When training is offered, notices shall be posted for all employees covered under this agreement for a period of not less than 14 calendar days. The notice shall contain all pertinent information, such as type of course, hours, duration and location of courses. Selection from applicants will be based on seniority among the applicants who meet the bona fide occupational requirements of the position. For the purposes of an employee being absent the provisions of Article 12.15 will apply provided that the training course has not yet commenced. Time spent in training shall be considered for all intents and purposes as time worked. Employees presently in the service of the Corporation will be considered for training before a person not already in the employ of the Corporation, unless otherwise mutually arranged. Those employees successfully completing the training for a given position will thereafter be considered qualified for that position unless there has been a significant material change to the duties of the position.

**16.5** Upon successful completion of necessary formal technical training, employees will be obligated to bid on any position for which they were trained on a senior may, junior must basis. The lock-in period will begin once an employee commences work on the position. If the training received was for more than four days and less than 15 days, the lock-in period will continue until they have completed 60 shifts worked. If the training period was for 15 days or more, the lock-in period will continue until they have completed 100 shifts worked.

If during the lock-in period the employee is the successful applicant to a permanent or temporary higher rated position, and is not permitted to immediately assume the position, he will be compensated at the higher rate until completion of the lock-in period, after which he will assume the higher rated position. The higher rate of pay will be discontinued if the employee is displaced, or the position is abolished during the lock-in period.

- **16.6** Upon successful completion of necessary formal technical training for the Locomotive Attendant position, the employee must remain on the position for at least 12 months of cumulative compensated service before being able to apply for a training on another position.
- **16.7** Authorized training expenses incurred by employees will be reimbursed by the Corporation.
- **16.8** Whenever an employee is expressly required to provide training on the job for one or more employees during his shift, he will be paid a premium of \$20.00 for each such shift.

#### ARTICLE 17 LEAVE OF ABSENCE & FREE TRANSPORTATION

- **17.1** Employees elected or appointed as full-time salaried National or Regional representatives paid by the Union shall, upon request, be granted leave of absence without pay and shall continue to accumulate seniority while so engaged. Such employees shall be considered as having accumulated cumulative compensated service under this Agreement for vacation purposes.
- **17.2** Employees shall be granted free transportation in accordance with pass regulations and leave of absence without pay to attend General Meetings, union conventions and union business upon the request of the Local Chairperson or designated National or Regional Representative of the Union.

- **17.3** Employees elected or appointed to serve on Union committees for investigation, consideration and adjustment of grievances shall, upon request, be granted free transportation in accordance with pass regulations and necessary leave of absence without pay.
- **17.4** Employees shall, upon request, be granted free transportation within their Area in accordance with pass regulations and leave of absence without pay to attend Union Meetings. Such leave of absence will be granted only when it will not interfere with the Corporation's business nor put the Corporation to additional expense.
- **17.5** Employees, at the discretion of the Corporation, may be granted a personal leave of absence without pay of up to four months, permission to be obtained in writing. The leave of absence may be extended by application in writing to the proper officer of the Corporation in ample time to receive permission or return to duty at the expiration of such leave.

Unless such extension of leave of absence is granted, or the employee provides a bona fide reason explaining why such return is prevented, a registered letter will be sent to the employee instructing him to report for an investigation in connection with the unauthorized leave of absence. If within a period of 30 calendar days from the date of the letter he fails to report for duty and investigation, he shall forfeit his seniority, his name shall be removed from the seniority list and his employment shall be terminated.

- **17.6** Leave of absence for educational purposes may be granted to employees in accordance with the Corporation's regulations. The designated National or Regional Representative of the Union will be informed when such leaves are granted. Such employees who return to the service between school terms, or prior to terminating the educational course for which leave of absence has been granted, will not be permitted to exercise their seniority.
- **17.7** Leave of absence under Article 17 shall not be granted for the purpose of engaging in work outside the Corporation's service, except in cases involving sickness, or when made the subject of mutual agreement between the proper officer of the Corporation and the designated National or Regional Representative of the Union.
- **17.8** The name of an employee on authorized leave of absence shall be continued on the seniority list for the group in which he has established seniority rights.
- **17.9** Employees on leave of absence for union business as provided under Article 17 for periods of less than a calendar month will contribute to the pension fund on the basis of constructive earnings as provided in VIA Rail Canada Inc. By-Law No. 20 effective March 1, 1982.

#### ARTICLE 18 SERVICE AWAY FROM HOME HEADQUARTERS

- **18.1** An employee who is regularly assigned to a position, the duties of which require him to be on the line from time to time, will be allowed necessary actual expenses while away from Headquarters. This will also apply to employees relieving on such positions.
- **18.2** A regularly assigned employee required to perform service away from the station at which regularly employed will be compensated in accordance with the schedule rules applicable at the point at which such service is performed for the time actually worked. Unless sleeping car accommodation is furnished or paid for by the Corporation such employee will be compensated at the hourly rate for the time occupied in travelling. The number of hours paid for will not be less than he would have earned on his regular assignment. Necessary actual expenses will be allowed while away from Headquarters when supported by receipts.

# ARTICLE 19 ATTENDING COURT

- **19.1** Employees who lose time by reason of being required to attend Court or Coroner's inquest or to appear as witnesses, in cases in which the Corporation is involved, will be paid for time so lost. If no time is lost, they will be paid for actual time held with a minimum of two hours at one and one-half times the hourly rate. Necessary actual expenses while away from home terminal will be allowed when supported by receipts.
- **19.2** Any fee or mileage accruing shall be assigned to the Corporation.

# ARTICLE 20 HELD FOR INVESTIGATION OR COMPANY BUSINESS

20.1 Employees held for Corporation's investigation and no responsibility is attached to them in connection with the matter under investigation (i.e., not subject to discipline) or on corporate business on the order of the proper officer will, if required to lose time by reason thereof, be paid for time lost. If no time is lost, they will be paid from the time required to report until actually released at one and one-half times the hourly rate, with a minimum of two hours. Necessary actual expenses will be allowed when supported by receipts.

Should the Corporation provide the employee with a Notice to Appear, the Notice will specify that the investigation is being held under this Article and a copy will be provided to the Local Chairperson.

#### ARTICLE 21 RELIEF WORK AND PRESERVATION OF RATES

- 21.1 An employee temporarily assigned for one hour or more, cumulative, in any one day, to a higher rated position, shall receive the higher rate while occupying such position, due regard being had to apprentice or graded rates. An employee temporarily assigned to a lower-rated position shall not have his rate reduced.
- 21.2 A "temporary assignment" contemplates the fulfilment of the duties and responsibilities of the position during the time occupied. Assisting a higher-rated employee due to a temporary increase in the volume of work does not constitute a temporary assignment.
- **21.3** Articles 21.1 and 21.2 shall not apply to a weekly rated employee who is filling a higher rated position through a higher rated employee being absent from duty with pay due to sickness or similar cause, other than vacation.
- 21.4 An employee engaged temporarily, or an employee temporarily promoted, on account of an employee being off duty without pay due to sickness or similar cause, or on vacation with pay, shall receive the rate applicable to the position on which employed, due regard being had to apprentice or graded rates.
- **21.5** The classifications and rates of pay for additional positions established on staffs covered by this Agreement shall be in conformity with classifications and rates of pay for positions of similar kind or class covered by this Agreement.
- **21.6** Established positions shall not be discontinued and new ones created covering relatively the same class of work for the purpose of reducing the rate of pay.

**21.7** No change shall be made in agreed classifications or basic rates of pay for individual positions unless warranted by changed conditions resulting in changes in the character of the duties or responsibilities. When changes in classifications and/or basic rates are proposed, or when it is considered that a position is improperly classified or rated, the work of the positions affected will be reviewed and compared with the duties and responsibilities of comparable positions by the Director, Human Resources, or designate, and the designated National or Regional Representative of the Union, with the object of reaching agreement on revised classifications and/or rates to maintain uniformity for positions on which the duties and responsibilities are relatively the same.

# ARTICLE 22 SERVICE LETTERS

- 22.1 A person, entering the service of the Corporation, will within 30 days from date of employment, have returned to him all service cards and letters of recommendation which had been taken up for inspection by the Corporation except for those addressed to or issued by the Corporation.
- 22.2 An employee who is dismissed or leaves the service of his own accord after giving due notice, will, upon request, be given the usual letter or reference and will be paid as soon as possible.

# ARTICLE 23 RATES OF PAY

23.1 An employee filling a clerical position (other than a VIAnet position) who has had less than 131 weeks' service with the Corporation will be paid according to the following:

0-26 weeks	-	25% less than the basic rate				
27-52 weeks	-	21% less than the basic rate				
53-78 weeks	-	17% less than the basic rate				
79-104 weeks	-	13% less than the basic rate				
105-130 weeks	-	9% less than the basic rate				
Thereafter the basic rate of the position will apply.						

**23.2** Employees other than clerical or VIAnet employees who have less than 26 weeks' service with the Corporation will be paid for those 26 weeks, 10% less than the basic rate. Thereafter the basic rate of the position will apply.

- **23.3** In applying Article 23, 40 cumulative hours worked will constitute one week. A general holiday for which the employee is paid will be considered a day worked. Service must be continuous. Service for rates of pay will be calculated based on the date of last entry into the Corporation's service.
- 23.4 An employee filling the position of Counter Sales Agent 1, Telephone Sales Agent, Tour Sales Agent, Rate & Refund Clerk, Special Traffic Clerk or Chief Passenger Clerk, who has had less than 131 weeks' service with the Corporation, will be paid in accordance with Appendix A.

(Appendix A will be modified accordingly).

**23.5** Newly hired employees will be paid the basic rate of the position they were hired for while in training.

Training hours for a new position will be paid the basic rate of pay for that position. However, the weeks/hours counted to reach the higher/next wage scale begin when the training is completed.

#### ARTICLE 24 DISCIPLINE AND GRIEVANCE PROCEDURE

- **24.1** An employee, who has completed his probationary period will not be disciplined or discharged without a fair and impartial investigation.
- 24.2 Investigation in connection with alleged irregularities will be held as quickly as possible. An employee may be held out of service for investigation (not exceeding three working days). He will be given at least 48 hours' notice in writing of the investigation and notified of the specific charges against him in writing. Where possible, investigations will be held during the employee's normal working hours. The Local Chairperson shall receive a copy of such notice. This shall not be construed to mean that a proper officer of the Corporation, who may be on the ground when the cause for investigation occurs, shall be prevented from making an immediate investigation. An employee may, if he so desires, have the assistance of one or two fellow employees, Local Chairperson, or authorized Committee person, at the investigation. The employee being investigated shall be furnished with a copy of his own statement if it is made a matter of record at the investigation. The decision will be rendered within 21 calendar days from the date the statement is taken from the employee being investigated. No discipline will apply if discipline is not assessed within 21 days from the date the statement is taken. An employee will not be held out of service pending the rendering of a decision, except in the case of a dismissible offence.

All evidence shall be made available to the designated local representative of the Union at the Corporation's office at least 48 hours in advance of the hearing, if he so desires. This does not preclude the Corporation from submitting, on an exceptional basis, additional evidence prior to the time of the hearing.

- 24.3 If a decision is considered unjust, an appeal may be made in writing within 21 calendar days in accordance with the Grievance Procedure. Such appeal shall set forth the grounds upon which it is made. The hearing on appeal shall be granted and a decision rendered as quickly as possible. On request, the designated National or Regional Representative of the Union shall be shown all evidence in the case.
- 24.4 Should an employee be exonerated he shall be paid at his regular rate of pay for any time lost (one day for each 24 hours), less any amount earned in other employment. If away from home he shall, on production of receipts, be reimbursed reasonable expenses for travelling to and from the investigation.
- 24.5 Any complaint raised by an employee concerning the interpretation, application or alleged violation of this Agreement shall be dealt with in the following manner: this shall also apply to an employee who believes that he has been unjustly dealt with.

#### Step 1

Within 21 calendar days from cause of grievance, the employee and/or the Local Chairperson, or his authorized committeeman, must present the grievance in writing to the immediate Supervisor who will give a decision within 21 calendar days of receipt of grievance.

## Step 2

Within 28 calendar days of receiving decision under Step 1, the Local Chairperson (or his designate) may appeal in writing to the Regional Head of the function or to the Head of the function at System Headquarters, as appropriate.

A decision will be rendered within 28 calendar days of receiving appeal. The appeal shall include a written statement of the grievance and where it concerns the interpretation or alleged violation of the Collective Agreement, the statement shall identify the Article and paragraph of the Article involved.

## Step 3

Within 60 calendar days of receiving decision under Step 2, the designated National or Regional Representative of the Union may appeal the decision to the Department Director, Labour Relations and Human Resources Services. The appeal shall include the written statement of the grievance which was presented at Step 2.

A decision will be rendered within 60 calendar days of receiving the appeal.

- **24.6** A grievance concerning the discipline of an employee may be processed commencing with Step 2 of the grievance procedure within 21 calendar days of the date the employee is notified of the discipline.
- **24.7** The settlement of a dispute shall not under any circumstances involve retroactive pay beyond a period of 60 calendar days prior to the date that such grievance was submitted at Step 1 of the Grievance Procedure.
- 24.8 Where any grievance is not progressed by the Union within the prescribed time limits, the grievance will be considered to have been dropped without precedent or prejudice. When the appropriate officer of the Corporation fails to render a decision with respect to a claim for unpaid wages within the prescribed time limits, the claim will be paid, but this will not constitute an interpretation of the Collective Agreement. Where a decision with respect to

a grievance other than one based on a claim for unpaid wages is not rendered by the appropriate officer of the Corporation within the prescribed time limits, it will be processed to the next step in the Grievance Procedure.

**24.9** The time limits provided under this Article may be extended by agreement between the Corporation officer and Union representative at any step.

# ARTICLE 25 FINAL SETTLEMENT OF DISPUTES

- **25.1** Provision shall be made in the following manner for the final and binding settlement, without stoppage of work, of differences or disputes, including personal grievances, which arise concerning the application or interpretation of this Agreement governing rates of pay and working conditions which cannot otherwise be disposed of between Officers of the Corporation and the Union.
- **25.2** A grievance concerning the interpretation or alleged violation of this Agreement or an appeal by an employee that he has been unjustly disciplined or discharged and which is not settled at Step 3 may be referred by either party to the Canadian Railway Office of Arbitration for final and

binding settlement without stoppage of work in accordance with the regulations of that Office.

- **25.3** The request for arbitration must be made in writing within 45 calendar days following the decision rendered at the last step by filing notice thereof with the Canadian Railway Office of Arbitration and on the same date by transmission of copy of such filed notice to the other party.
- **25.4** The time limits provided under this Article may be extended by agreement between the designated National or Regional Representative of the Union and the Department Director, Labour Relations of the Corporation.

## ARTICLE 26 HEALTH AND WELFARE

**26.1** Health and Welfare benefits will be provided in accordance with the amendments as set forth in the Memorandum of Agreement dated July 12, 2022.

# ARTICLE 27 GENERAL

- 27.1 Employees will be paid by direct deposit, electronic transfer funds, every other Thursday during their regular work hours. When a holiday falls on a Thursday which is a pay-day, employees will be paid on the preceding Wednesday.
- 27.2 (a) The daily rate of pay shall be the weekly rate divided by five.
  - (b) Weekly and daily rates will be calculated to the nearest whole cent figure.
  - (c) In any situation where an employee's regular assignment is other than on a five-day week basis, the formula specified in Article 27.2

     (a) will be adjusted accordingly.
  - (d) The hourly rate of pay will be computed to the nearest tenth of a cent by dividing the weekly rate of pay by 40.
- (a) When an employee is short paid \$120.00 or more, on request of the employee, a voucher will be issued equal to the outstanding amount within 72 hours following the scheduled pay day, excluding Saturdays, Sundays, and statutory holidays.

- (b) In the event the employee receives payment for 80 hours within the pay period, the voucher for the outstanding amount will be issued in the following pay period.
- (c) The sum specified in paragraph (a) herein will be adjusted to reflect future general wage increases.
- 27.4 An employee transferred by direction of the Corporation to positions which necessitate a change of residence will receive free transportation for himself, dependent members of his family and the cost of a truck rental, gasoline and insurance, in line with the Corporation's regulations. Such employee will be compensated for time lost up to a maximum of three days, unless otherwise mutually arranged.
- 27.5 An employee exercising seniority rights to a position which necessitates a change of residence will receive free transportation for himself, dependent members of his family and the cost of a truck rental, gasoline, and insurance, in line with the Corporation's regulations. Such free transportation under these circumstances will not be allowed more than once in a 12-month period. A free billing order to cover transportation of furniture and effects must be applied for and issued before shipment is arranged by employees.
- 27.6 At points or in departments where there are employees, it will be permissible for notices of interest to said employees to be posted. The notice board shall be supplied by the employees and shall be in keeping with the general furnishings.
- 27.7 Where practicable, reasonable efforts will be made to ensure that premises will be heated, lighted and ventilated; where there is actual need, suitable accommodation will be provided in which the employee may eat.
- **27.8** Where an automobile mileage allowance is paid, such allowance shall be 40 cents per kilometer.
- **27.9** All overtime earned shall be shown as a separate item on the pay cheques of employees.
- 27.10 Upon reasonable advance request to his immediate supervisor, an employee shall be allowed to view his personal file in the presence of the supervisor and may request the removal of irrelevant medical information from it. The Chairperson or authorized committee person may also be present if the employee so wishes.
- 27.11 Any employee responsible for Corporation funds is also responsible for overages and shortages but will not be required to reimburse shortages unless the Corporation establishes intent to defraud.

- 27.12 Monitoring of employee performance in the context of quality control or training programs, will not be used in a discriminatory fashion nor for disciplinary purposes.
- 27.13 Employees required to punch clocks in and out and make service cards on their own time will be allowed a bonus of one minute per hour. This article is not applicable to hourly rates employees in the Equipment Maintenance Facilities who have had this allowance incorporated into their basic rate of pay.
- 27.14 The Corporation will undertake the responsibility for the translation of this Agreement as may be required from time to time and will absorb the cost of such translating. This will include the cost of translating updated pages.

The parties agree once language verification and translation have been completed, the Corporation will immediately post Collective Agreements No.1 Letters of Understanding, Benefits Plans, Health and Safety Agreement, LTD, ESIMA, etc. on its intranet (IVIA) site. The Union will have the opportunity to review the translated agreement for accuracy and completeness prior to posting on IVIA.

The parties agree once language verification and translation have been completed, the Corporation will immediately send an updated copy of the Collective Agreements No.1, Letters of Understanding, Benefit Plans, Health and Safety Agreement, LTD, ESIMA, etc. in electronic format and sufficient printed booklets in (8.5 x 11) format to the President of Council 4000 and workplace.

- 27.15 Employees will be provided with free transportation privileges in accordance with the Corporation's policy. Employees on laid-off status and eligible dependents will be granted pass privileges in accordance with the Corporation's regulations, for the period that they remain employees of the Corporation.
- 27.16 The following types of work shall be performed by employees governed by this agreement:
  - (1) unloading fuel and lubricating oil from railway tank cars to storage tanks and vice versa, also the manning of railway fuel pumps during any operation from any vehicle. At smaller terminals where insufficient work is available to justify the employment of classified labourer, the performance of such work shall be at the discretion of the company;

- (2) fueling of any type of locomotive equipment using fuel oil or the responsibility of checking fuel oil prior to the dispatch of such equipment. At points where there is insufficient work of the above nature to justify a full-time employee, the performance of such work shall be at the discretion of the company;
- (3) drying sand and filling sand boxes on diesel and electrical units;
- (4) filling of water tanks for steam generators of Diesel Units and Steam Generator Cars;
- (5) mixing of compound and supplying same to Diesel Units;
- (6) when the company considers it necessary, responsibility for watching Diesel Units, engines of which are required to be kept idling, and movement of Diesel Units on shop tracks at subsidiary stations;
- (7) transcribing inspection records and technical data into records and files;
- (8) filling lubricators inside and outside, cleaning, fueling, and placing supplies on locomotives and roundhouses to be performed by classified labourers.
- (9) Locomotive Attendants are responsible for the movement of locomotives and train consists on shop tracks within yard limits and at subsidiary stations. Accompanying locomotives and train consists on shop tracks within yard limits and where necessary moving locomotives and train consists beyond the recognized shop track switch and yard limits.
- 27.17 The Corporation and the Union agree that there shall be no discrimination exercised or practiced with respect to any employee governed by this agreement in the matters of hiring, training, promotion, lay-off, transfer, recall, discipline or discharge for any reason, except bona fide occupational requirements.
- 27.18 It is agreed that there shall be no discrimination or harassment towards an employee by management personnel or by another employee based on race, national or ethnic origin, color, religion, age, sex, sexual orientation, gender identity and expression, marital status, family status, disability, or conviction for which a pardon has been granted.

The prohibited grounds of discrimination and harassment are defined and interpreted in accordance with the Canadian Human RightsAct.

It is further agreed that management personnel instructing an employee to perform duties or directing their work is not discrimination or harassment unless based on one or more of the prohibited grounds.

- **27.19** Any Union employee experiencing personal problems shall be entitled to utilize the services of the VIA E.A.P. Plan, as the provisions of that Plan dictate.
- 27.20 Upon request from the Union, the Corporation will provide one designated Representative of the Union in September of each year with a list of employees governed by this Agreement, which shall include the employees' home address and telephone number. The authorized Union Representative must make this request in writing.

## ARTICLE 28 WAGE RATES FOR NEW JOBS

- **28.1** When a bona fide new job or position is to be established which cannot be properly placed in an existing classification by mutual agreement management will establish a classification and rate on a temporary basis.
- **28.2** Written notification of the temporary rate and classification will be furnished to the designated National or Regional Representative of the Union.
- **28.3** The new rate and classification shall be considered temporary for a period of 60 calendar days following the date of notification to the designated National or Regional Representative of the Union. During this period (but not thereafter) the designated National or Regional Representative of the Union may request the Corporation to negotiate the rate for the classification.

The negotiated rate, if higher than the temporary rate, shall be applied retroactively to the date of the establishment of the temporary classification and rate, except as otherwise mutually agreed. If no request has been made by the Union to negotiate the rate within the 60-calendar day period, or if no grievance is filed within 60 calendar days from the date of notification to the designated National or Regional Representative of the Union, or upon completion of negotiations, as the case may be, the temporary classification and rate shall become a part of the wage scale.

- **28.4** If the Corporation and the Union are unable to agree on a classification, and rate for the new job, the disputed rate and/or classification may be treated as a grievance. The grievance may be taken up at Step No. 3 of the grievance procedure and if it is not resolved, it may be referred to arbitration under Article 25.
- 28.5 It is specifically agreed that no arbitrator shall have the authority to alter or modify the existing classifications or wage rates, but he shall have the authority, subject to the provisions of this Agreement, to determine whether or not a new classification or wage rate has been set properly within the framework of the Corporation established classification and rate setting procedure.

# ARTICLE 29 BEREAVEMENT LEAVE

- **29.1** Provided an employee has at least three months' cumulative compensated service, they will be granted bereavement leave (excluding rest days and annual vacation) without loss of pay as follows:
  - (a) **a paid leave of a maximum duration of 5 days and an unpaid leave for a maximum duration of 5 days each time there is a death in your family,** upon the death of the employee's parents, child (including stillborn child) or spouse or common law partner that may be taken during the period that begins on the day on which the death occurs and ends six weeks after the latest of the days on which any funeral, burial or memorial service of that immediate family member occurs.
  - (b) three (3) days and an unpaid leave for a maximum duration of 7 days, upon the death of an immediate family member that may be taken during the period that begins on the day on which the death occurs and ends six weeks after the latest of the days on which any funeral, burial or memorial service of that immediate family member occurs. Furthermore, the employee will be entitled to two (2) additional days without compensation for the same lost.

For the purpose of paragraph (b), Immediate Family means, in respect of an employee,

- (a) the spouse or common-law partner of the employee's father or mother;
- (b) the children of the employee's spouse or common-law partner;
- (c) the employee's grandchildren;
- (d) the employee's brothers and sisters;
- (e) the grandfather and grandmother of the employee;
- (f) the father and mother of the spouse or common-law partner of the employee and the spouse or common-law partner of the father or mother;

and

- (g) any relative of the employee who resides permanently with the employee or with whom the employee permanently resides.
- (h) the brother or sister-in-law
- (i) Foster Child

The leave of absence may be taken in one or two periods. The employer may require that any period of leave be of not less than one day's duration.

- **NOTE:** In the application of this Article "employee's spouse" means the person who is legally married to the employee and who is residing with or supported by the employee, provided that, if there is no legally married spouse, common-law partner means a person who has been cohabiting with an individual in a conjugal relationship for at least one year, or who had been so cohabiting with the individual for at least one year immediately before the individual's death.
- 29.2 An employee shall advise in writing if they wish to postpone and/or have a single split of their bereavement leave to enable them to attend a memorial service or event that takes place at a later date within the following twelve months. The employee will be entitled to no more than the leave from work and payment of their regular wages and guarantee that they would have been entitled to on the occasion of death in accordance with Articles 29.1 and Appendix H.
- 29.3 An employee on vacation who suffers the loss of a family member as set out within this Article shall have the right to terminate (temporarily) his/her vacation and be granted bereavement leave. The employee shall immediately inform the Corporate office in charge and will continue his vacation if within his scheduled dates. If the remaining vacation falls outside the employee's scheduled dates, such vacation will be rescheduled

# ARTICLE 30 JOB SECURITY, TECHNOLOGICAL, OPERATIONAL AND ORGANIZATIONAL CHANGES

**30.1** The provisions of the Supplemental Agreement governing Employment Security and Income Maintenance dated June 1, 1998, will apply to employees covered by this Agreement.

**30.2** Whenever the Corporation is contemplating a closure of any shop, facility, installation, or office where employees who are subject to this Agreement are employed and which is currently in operation, the Corporation must give the Union at least 90 days written notice of its intent. In addition, the

Corporation must meet with the Union within five days of receiving a written request from the Union for the meeting, to consult about:

- a) possible alternatives to the closure;
- b) plans for the future of the affected employees.

Should the Corporation fail to consult with the Union in the above manner, its written notice shall be deemed void, and the closure shall be unable to take place.

# ARTICLE 31 JURY DUTY

- **31.1** An employee who is summoned for jury duty and is required to lose time from his assignment as a result thereof, shall be paid for actual time lost with a maximum of one basic day's pay at the straight time rate of his position for each day lost, less the amount allowed him for jury duty for each such day excluding allowances paid by the court for meals, lodging or transportation, subject to the following requirements and limitations:
  - (a) An employee must furnish the Corporation with a statement from the court of jury allowances paid and the days on which jury duty was performed.
  - (b) The number of working days for which jury duty pay shall be paid is limited to a maximum of 120 days in any calendar year.
  - (c) No jury duty pay will be allowed for any day for which the employee is entitled to vacation or general holiday pay. If an employee is to be on jury duty during any period when he was scheduled to be on annual vacation, the employee may request to have his annual vacation rescheduled to a later date, and the Corporation must either reschedule that vacation before the end of the calendar year when the vacation was originally scheduled or pay the employee in lieu of the rescheduled vacation at straight time rates at the Corporation's discretion.

## ARTICLE 32 SHIFT DIFFERENTIALS

**32.1** Employees whose regularly assigned shifts commence between 14:00 and 21:59 hours (evenings) or between 22:00 and 05:59 hours (nights) shall receive a shift differential as follows:

Evening:	\$1.50
Night:	\$2.00

Overtime shall not be calculated on the shift differential, nor shall the shift differential be paid for paid absence from duty such as vacations, general holidays, etc.

#### ARTICLE 33 PENSION PLAN WIND UP

**33.1** In the event that the employee's pension plan is wound up, the pension fund shall be used for the sole purpose of providing benefits to Plan members and survivors. In the event that the Revenue Canada maximum pensions are reached for all Plan members, any surplus then remaining in the pension fund shall revert to the Corporation.

#### ARTICLE 34 LIFE INSURANCE UPON RETIREMENT

34.1 An employee who retires from the service of the Corporation subsequent to August 1, 2001, will, provided he is 55 years of age or over and has not less than 10 years' cumulative compensated service, be entitled to the sum of \$8,000.00, payable to his estate upon his death.

## ARTICLE 35 DENTAL PLAN

**35.1** Dental Plan benefits will be provided in accordance with the terms set forth in the Memorandum of Agreement dated July 12, 2022.

# ARTICLE 36 EXTENDED HEALTH CARE PLAN

- **36.1** Extended Health Care benefits will be provided in accordance with the terms set forth in the Memorandum of Agreement dated July 12, 2022.
  - **NOTE:** Short Term Disability R.A.A.Q. and similar provincial insurance payments are to be deducted from short term disability payments.

## ARTICLE 37 PAID MATERNITY LEAVE

- **37.1** During her maternity leave, an employee is entitled to maternity leave payments in an amount equal to 85% of her weekly base pay up to a maximum of 20 weeks. If during that period the employee is entitled to receive any benefits, like Employment Insurance maternity benefits, the amount of such benefits will reduce the amount to be paid by the Corporation.
- **37.2** An employee who is the biological parent of a newborn child or an employee who commences legal proceedings to adopt a child and the child comes under his or her care will be entitled to a leave of absence without pay in accordance with the terms and conditions of the Canada Labour Code Part III.
- **37.3** The maternity leave and parental leave combined cannot exceed 78 weeks. Such leave will not affect the employee's seniority. Employees must buy- back their pension for that year.
- **37.4** Upon the birth or the adoption of an child of an employee's partner, the employee shall receive one day of paid leave.

#### ARTICLE 38 UNIFORMS

- **38.1** Effective January 1, 1999, employees in service who are required to wear a dress uniform will receive the uniform and all subsequent issues and necessary replacements, free of charge. Employees who leave the service of the Corporation will return all items of their uniform to the Corporation.
- **38.2** Each employee required to wear a dress uniform will be responsible for cleaning, maintaining, and storing the uniform, and will, while actually working on a position which requires the wearing of a dress uniform receive:
  - (i) a uniform cleaning and maintenance allowance of \$33.75 per month.

- **38.3** Employees required to wear protective clothing (such as rubber gloves, aprons) will receive such clothing free of charge. Such protective clothing will be cleaned and maintained by the Corporation.
- 38.4 An employee, who is required by the Corporation to wear safety footwear and was in service at the beginning of the calendar year, has rendered compensated service during the year, and holds employment relationship, will be entitled to a safety footwear allowance of 150\$ payable in the second pay period of September each year. When an employee purchases new safety footwear, he or she must report to his or her immediate supervisor with the footwear and proof of purchase indicating that such footwear is in compliance with the Canada Occupational Safety & Health Regulations (C.S.A. approved).
- 38.5 Non-clerical employees assigned to work in the maintenance centers (including employees working in the Employee Service Centre at the Toronto Maintenance Centre) who do not normally come into contact with customers, have rendered compensated service during the year and hold an employment relationship with the Corporation will receive a coverall allowance of \$120 dollars payable in the 20th pay period of each year
- **38.6** Employees holding permanent assignments that regularly require them to work outdoors during winter months will be provided with suitable outerwear within 12 months of signing and every four (4) years thereafter. Those awarded these positions in the future will be provided with the appropriate outerwear as soon as possible.

The outerwear will consist of a parka and pants, one-piece winter suit or rain gear depending on the work location.

At the discretion of the Corporation, it may, in exceptional circumstances, replace outerwear earlier than four (4) years that has been destroyed due to ordinary wear and tear.

# ARTICLE 39 SAFETY, HEALTH AND ENVIRONMENT

**39.1** The Corporation and the Union reached an agreement on February 22, 1999, regarding Safety, Health and Environment. The agreement does not form part of the Collective Agreement.

#### ARTICLE 40 DURATION OF COLLECTIVE AGREEMENT

40.1 This Collective Agreement is in full settlement of all issues raised by either party on or subsequent to January 17, 2022. It is effective January 1, 2022, unless otherwise stipulated in the Memorandum of Agreement dated July 12, 2022 and supersedes all previous agreements, rulings or interpretations which are in conflict therewith. It will remain in effect until December 31, 2024, and thereafter, until revised or superseded. Four months' notice shall be given by either party of its desire to revise or supersede this Agreement and may be served at any time subsequent to August 31, 2024.

Signed at Montreal, Quebec, this \_\_\_\_\_ of \_\_\_\_\_, 2022.

FOR THE UNION:

FOR THE CORPORATION:

Dave Kissack President, UNIFOR Council 4000 Karine Chapados Specialist Advisor, Employee relations VIA Rail Canada

# APPENDICES

# **APPENDIX A**

#### WAGE SCALE

NON-CLERICAL POSITIONS						
CLASSIFICATION		Wage Rate (\$)				
POSITION	LEVEL	WEEKS	2022	2023	2024	
Red Cap	А	0-26 27 & over	\$909.60 \$1010.80	\$941.35 \$1046.06	\$964.88 \$1072.22	
Red Cap Captain	В	0-26 27 & over	\$970.64 \$1078.25	\$1004.61 \$1115.99	\$1029.73 \$1143.89	
Janitor	С	0-26 27 & over	\$1068.50 \$1187.15	\$1105.90. \$1228.70	\$1133.55 \$1259.42	
Stock Attendant	C/D	0-26 27 & over	\$1111.82 \$1235,36	\$1150.73 \$1278.60	\$1179.50 \$1310.57	
Station Service Attendant	D	0-26 27 & over	\$1130.96 \$1255.60	\$1170.54 \$1299.55	\$1199.81 \$1332.04	
Stock Checker ticket Seller/Examinor Stores Attendant	E	0-26 27 & over	\$1157.02 \$1285.83	\$1197.52 \$1330.84	\$1227.46 \$1364.11	
Senior Station Services Agent	F	0-26 27 & over	\$1184.58 \$1315.86	\$1226.04 \$1361.92	\$1256.70 \$1395.97	
Senior Stockchecker Senior Stores Attendant	G	0-26 27 & over	\$1211.14 \$1345.76	\$1253.53 \$1392.86	\$1284.87 \$1427.68	

NON-CLERICAL POSITIONS							
CLASSIFICATION	v	Wage Rate (\$)					
Positions	Weeks	2022	2023	2024			
Classified Labourer	0-26	\$27.02	\$27.96	\$28.66			
	27 & over	\$30.03	\$31.08	\$31.85			
Leading Hand Classified	0-26	\$28.02	\$29.00	\$29.73			
Labourer	27 & over	\$31.13	\$32.22	\$33.03			
Fork Lift/Tow Motor	0-26	\$28.33	\$29.32	\$30.05			
Operator	27 & over	\$31.46	\$32.56	\$33.38			
Truck Driver 5 Ton Pettibone Crane Operator	0-26 27 & over	\$29.28 \$32.56	\$30.30 \$33.70	\$31.06 \$34.54			
Locomotive Attendant	0-26	\$34.08	\$35.28	\$36.11			
	27 & over	\$37.63	\$38.94	\$39.87			

CLERICAL POSITIONS							
CLASSIFICATION	Wage Rate (\$)						
Position	Position Level Weeks			2023	2024		
Office Service Clerk Data Entry Clerk	С	0-26 27-52 53-78 79-104 105-130 131 & over	\$911.25 \$960.05 \$1007.74 \$1057.53 \$1105.64 \$1214.71	\$943.14 \$993.65 \$1043.01 \$1094.55 \$1144.34 \$1257.22	\$966.72 \$1018.49 \$1069.08 \$1121.91 \$1172.95 \$1288.65		
Word Processor Operator	D	0-26 27-52 53-78 79-104 105-130 131 & over	\$942.33 \$992.12 \$1042.34 \$1092.56 \$1142.35 \$1255.60	\$975.31 \$1026.85 \$1078.82 \$1130.80 \$1182.34 \$1299.55	\$999.69 \$1052.52 \$1105.79 \$1159.07 \$1211.89 \$1332.04		
Administration Support Clerk Work Processor Specialist	E	0-26 27-52 53-78 79-104 105-130 131 & over	\$964.69 \$1016.18 \$1067.24 \$1118.72 \$1169.36 \$1285.83	\$998.46 \$1051.74 \$1104.59 \$1157.88 \$1210.29 \$1330.84	\$1023.42 \$1078.04 \$1132.21 \$1186.62 \$1240.55 \$1364.11		
Administration Clerk Work Order Copy Clerk	F	0-26 27-52 53-78 79-104 105-130 131 & over	\$987.48 \$1039.39 \$1092.56 \$1144.46 \$1197.64 \$1315.86	\$1022.04 \$1075.76 \$1130.80 \$1184.52 \$1239.55 \$1361.92	\$1047.59 \$1102.66 \$1159.07 \$1214.13 \$1270.54 \$1396.96		

CLERICAL POSITIONS							
CLASSIFICATION	Wage Rate (\$)						
Position	Level	Weeks	2022	2023	2024		
Senior Administration Clerk Accounting Clerk Payroll Clerk General Operations Clerk Shipper/Receiver	G	0-26 27-52 53-78 79-104 105-130 131 & plus	\$1009.00 \$1062.60 \$1117.03 \$1170.21 \$1224.64 \$1345.76	\$1044.32 \$1099.79 \$1156.13 \$1211.16 \$1267.51 \$1392.86	\$1070.42 \$1127.28 \$1185.03 \$1241.44 \$1299.19 \$1427.68		
Inventory Control Clerk Senior Accounting Clerk Senior Payroll Clerk Communications Operator	н	0-26 27-52 53-78 79-104 105-130 131 & plus	\$1035.21 \$1090.45 \$1144.46 \$1200.17 \$1255.45 \$1379.56	\$1071.44 \$1128.61 \$1184.52 \$1242.17 \$1299.39 \$1427.84	\$1098.23 \$1156.83 \$1214.13 \$1273.23 \$1331.88 \$1463.54		
Control Clerk Shared Services Clerk Senior Computer Operator VIAnet Text Clerk Senior Finance Accounting Clerk	I	0-26 27-52 53-78 79-104 105-130 131 & plus	\$1058.88 \$1114.92 \$1171.05 \$1228.44 \$1284.99 \$1412.01	\$1095.94 \$1153.95 \$1212.04 \$1271.44 \$1329.96 \$1461.43	\$1123.34 \$1182.79 \$1242.34 \$1303.22 \$1363.21 \$1497.97		

CLERICAL POSITIONS						
CLASSIFICATION			Wage Rate (\$)			
Position	Level	Week	2022	2023	2024	
Senior Employee Service Centre Attendant Senior Control Clerk	J	0-26 27-52 53-78 79-104 105-130 131 & over	\$1086.43 \$1143.62 \$1202.28 \$1260.09 \$1318.00 \$1448.40	\$1124.46 \$1183.65 \$1244.36 \$1304.20 \$1364.03 \$1499.13	\$1152.40 \$1213.20 \$1275.60 \$1336.80 \$1398.00 \$1536.80	

VIANET RELATED POSITIONS							
CLASSIFICATION	CLASSIFICATION						
Positions	Positions Weeks			2024			
Counter Sales Agent Telephone Sales Agent Rate & Refund Agent Special Traffic Agent Internet Agent	0-13 14-26 27-39 40-52 53-78 79-104 105-130 131& more	\$976.67 \$1028.84 \$1079.05 \$1131.38 \$1182.02 \$1232.66 \$1285.41 \$1412.01	\$1010.85 \$1064.85 \$1116.82 \$1170.98 \$1223.39 \$1275.81 \$1330.40 \$1461.43	\$1036.12 \$1091.47 \$1144.74 \$1200.25 \$1253.98 \$1307.70 \$1363.66 \$1497.97			
Senior Counter Sales Agent Senior Telephone Sales Agent Senior Internet Agent Senior Trade Desk Agent	Sales Agent Senior\$1448.43Telephone Sales\$1448.43Agent Senior\$1499.13Internet Agent\$16000000000000000000000000000000000000						
NOTE 1:       Employees presently holding a VIANet related position will be allowed a maximum of 30 working days credit in any calendar year towards qualifying for a higher step rate for the following reasons: <ul> <li>a) authorized leaves for absence such as :</li> <li>(i) bona fide illness, injury;</li> <li>(ii) maternity;</li> <li>(iii) to attend committee meetings;</li> <li>(iv) jury duty;</li> <li>(v) called to court as a witness for the Corporation;</li> <li>(vi) parental leave;</li> <li>(vii) leave for union business.</li> </ul> <li>For the purpose of this Agreement, 21 working days will constitute one month.</li> <li>NOTE 2: Service must continuous. Service for rates of pay will be calculated based on the date of last entry in the Corporation's service.</li>							

#### **APPENDIX B**

#### **REGULATIONS GOVERNING**

#### THE APPLICATION OF

#### SUPPLEMENTAL SENIORITY LISTS

#### PRIOR TO JUNE 14,1995:

A supplemental seniority list is established on each Region for employees promoted on or after December 29, 1978, to a permanent non-schedule, official or excluded position with the Corporation, or its subsidiaries.

As provided in Article 11.9 herein:

- a) The names of employees will not be removed from the existing seniority list and placed on the supplemental seniority list until they stop accumulating seniority.
- Employees promoted to a permanent non-schedule, official or excluded position with the Corporation, or its subsidiaries on or after December 29, 1978, shall, after a period of 2 consecutive years from the date on which promoted, have their names placed on the supplemental seniority list in seniority order with accumulated seniority up to the date of promotion.
- c) Employees covered by the provisions of Item 2(b) above who are released from excepted employment will have their names deleted from the supplemental seniority list and their names will be placed on the Regional seniority list in accordance with their seniority which they accumulated up to the date of promotion.
- d) Employees promoted to a permanent non-schedule, official or excluded position with the Corporation or its subsidiaries on or after December 29, 1978, shall after a period of 5 consecutive years from the date on which promoted have their names removed from the supplemental seniority list and such employees will at such time forfeit all seniority rights under Agreement 1.
  - **NOTE:** The above regulations are extracted from a Memorandum of Agreement signed on January 29, 1982, between the Canadian Brotherhood of Railway, Transport and General Workers Union of Canada and VIA Rail Canada Inc.

# EFFECTIVE JUNE 14, 1995:

A supplemental seniority list is established on each Region for employees promoted on or after June 14, 1995, to a non-schedule, official or excepted position with the Corporation.

As provided in Article 11.9 herein:

- a) Employees promoted to a non-schedule, official or excepted position with the Corporation on or after June 14, 1995, will continue to accumulate seniority for a period of 12 consecutive months from the date on which promoted.
- b) Employees shall thereafter, have their names placed on the supplemental seniority list in seniority order.
- c) Employees promoted to a non-schedule, official or excepted position who are set back to a position covered by this Collective Agreement for a period of less than 3 months, shall have such time considered as part of the 12 consecutive months.
- d) Employees who are released from excepted employment will have their names deleted from the supplemental seniority list and their names will be placed on the regional seniority list in accordance with the seniority which they accumulated up to 12 months after the promotion date.

## **APPENDIX C**

### LETTER ON CONTRACTING OUT

May 14, 1987

Mr. T. McGrath National Vice-President C.B.R.T. & G.W. 2300 Carling Avenue Ottawa, Ontario K2B 7G1

Dear Mr. McGrath:

This has reference to the award of the Arbitrator, the Honourable Emmet M. Hall, dated December 9, 1974, concerning the contracting out of work.

In accordance with the provisions as set out on Page 49 of the above-mentioned award, it is agreed that work presently and normally performed by employees represented by the Union will not be contracted out except:

- 1) when technical or managerial skills are not available from within the Railway; or
- 2) where sufficient employees, qualified to perform the work, are not available from the active or laid-off employees; or
- 3) when essential equipment or facilities are not available and cannot be made available from Railway-owned property at the time and place required; or
- 4) where the nature or volume of work is such that it does not justify the capital or operating expenditure involved; or
- 5) the required time of completion of the work cannot be met with the skills, personnel or equipment available on the property; or
- 6) where the nature or volume of the work is such that undesirable fluctuations in employment would automatically result.

The conditions set forth will not apply in emergencies, to items normally obtained from manufacturers or suppliers, nor to the performance of warranty work.

It is further agreed that at a mutually convenient time at the beginning of each year and, in any event, no later than January 31, of each year, representatives of the Union will meet with the designated officers to discuss the Corporation's plans with respect to contracting out of work for that year. In the event Union representatives are unavailable for such meetings, such unavailability will not delay implementation of the Corporation plans with respect to contracting out of work for that year.

In addition, the Corporation will advise the Union representatives involved in writing, as far in advance as practicable, of its intention to contract out work which would have a material and adverse effect on employees. Except in case of emergency, such notice will be no less than 30 days.

Such advice will contain a description of the work to be contracted out; the anticipated duration; the reasons for contracting out and, if possible, the date the contract is to commence. If the Union designated National or Regional Representative, or equivalent, requests a meeting to discuss matters relating to the contracting out of work specified in the above notice, the appropriate Corporation representative will promptly meet with him for that purpose.

Should a designated National or Regional Representative or equivalent, request information respecting contracting out which has not been covered by a notice of intent, it will be supplied to him promptly. If he requests a meeting to discuss such contracting out, it will be arranged at a mutually acceptable time and place.

Where the Union contends that the Corporation has contracted out work contrary to the foregoing, the Union may progress a grievance by using the grievance procedure which would apply if this were a grievance under the Collective Agreement. Such grievance shall commence at the second step, the Union officer submitting the facts on which the Union relies to support its contention. Any such grievance must be submitted within 30 days from the alleged non-compliance.

Yours truly,

(signed A.D. Andrew) A.D. Andrew Director, Labour Relations

## **APPENDIX D**

#### BILINGUALISM

A national approach will be utilized for identifying and providing the bilingual requirements of the Corporation in order to serve the travelling public. Representatives of the Union and the Corporation will meet to discuss the bilingual requirements for the System before any changes are implemented.

Both parties recognize that there are already many employees with bilingual skills. Where bilingual employees are already available in the positions required, formal designation would be unnecessary. Accordingly, attention will be focused on identifying specific positions only when the status quo has failed to fulfill the needs.

If the Union disagrees with the designation of any specific position to satisfy the needs of the travelling public or to comply with the Official Languages Act, a grievance may be commenced at Step 3 of the Grievance Procedure within 60 days of receiving the Corporation's decision to designate said position. If the grievance is not regulated at Step 3, the matter may be referred to arbitration in an expedited manner.

A language training bulletin will be posted twice per year for a 15-day period, inviting applications from employees desiring to qualify in the bilingual requirements for positions covered by this Agreement. Unilingual employees will be given language training in seniority order, or as mutually arranged. Employees having completed the home study language training will be compensated 40 hours pay at the classification rate of pay last worked immediately prior to commencing training for each module completed.

Employees who would prefer to study the second language at community colleges or local educational institutions, shall, with prior approval of the Corporation, receive full reimbursement for tuition fees, and required learning materials, upon presentation of proof of successful completion of the course.

After a position has been designated bilingual, efforts to staff it with a bilingual employee will be made if and when the regularly assigned position becomes vacant. Bilingual employees who are working on other positions will not be forced to fill bilingual positions, unless it is necessary to staff a customer contact position with a bilingual employee to comply with the Official Languages Act.

In any event, no employees will be compelled to accept work outside their home stations or terminals, or off their regularly assigned train solely because of bilingualism. Unilingual employees will not be laid-off or forced to take the spare board solely because they are not bilingual. If in the case of a reduction of staff, a unilingual employee would otherwise have been laid-off solely because he is not bilingual, he would in that case be permitted to displace a junior employee from a designated bilingual position. A unilingual employee who by reason only of not being bilingual, is unable to hold a position at his existing rate of pay, will be assured maintenance of earnings, until he is trained for a bilingual position.

If at the time of the staff reduction (for whatever reason) a unilingual employee is unable to hold a position at his existing rate of pay, solely because he is not bilingual, and is required to take a lower-rated position, he will be assured maintenance of earnings at his existing rate of pay, until he is trained for a bilingual position. If he is subsequently able to hold a position at least equal to the level of his maintenance of earnings, the maintenance of earnings will thereupon be eliminated. However, if he subsequently again is compelled to take a lower-rated position, solely because he is not bilingual, he will again be assured maintenance of earnings, and so on, until he is trained for a bilingual position.

Unilingual employees who are holding designated jobs on the date this agreement comes into effect, who are presently rated at the "C" level in the second language, and whose test results indicate that they could achieve bilingual status with a minimum exposure to immersion training, will be offered such training, with salary protection for up to four (4) weeks salary at their basic rate of pay, if the employee requests it.

### APPENDIX E

### LETTER RE PART-TIME EMPLOYEES

May 14, 1987

Our File: 25-8315-1-1

Mr. T. McGrath National Vice-President C.B.R.T. & G.W. 2300 Carling Avenue Ottawa, Ontario K2B 7G1

Dear Mr. McGrath:

The Memorandum of Settlement dated May 14, 1987, relating to Collective Agreement No. 1 contains some new provisions dealing with the utilization of part-time employees. It is understood that this category will include what used to be known as "extra and unassigned" employees and employees doing "spare and relief" work. This will also confirm that the total of part-time employees will be adjusted in accordance with the new provisions.

If the foregoing also reflects your understanding, would you please so indicate by signing below.

Yours truly,

(signed A.D. Andrew) .D. Andrew Director, Labour Relations ADA/pg

I Concur:

(signed T. McGrath) T. McGrath

### APPENDIX F

#### LETTER ON PAYMENT DURING TRAINING

June 26, 2010

Ms. Heather Grant Secretary Treasurer National Council 4000 CAW Canada

Dear Ms. Grant

During the negotiation of the Memorandum of Settlement for Collective Agreement No. 1 the Corporation raised the issue of payment during training.

This will confirm our understanding that employees receiving training prior to being appointed to the position for which trained will be paid at the rate of their regular position during the related training period. Following qualification, the employee will be reimbursed the difference between their regular rate and the rate for the position for which they recently qualified. This retroactive payment will be made during the next pay period following completion of the training.

If the foregoing clearly reflects our understanding, would you please so indicate by signing below.

Yours truly,

I concur,

<u>(Signed E.J. Houlihan)</u> E.J. Houlihan Director, Labour Relations <u>(Signed Heather Grant)</u> Heather Grant Secretary Treasurer

# APPENDIX G

### LEFT BLANK BY THE PARTIES

## **Expedited Mediation/Arbitration of Grievances**

BETWEEN: Unifor National Council 4000 Canada (hereinafter referred to as the "Union")

AND: VIA Rail Canada Inc. (hereinafter referred to as the "Corporation")

**WHEREAS** the Union expressed a concern during negotiations that grievances were not being heard on a timely basis due to backlogs at CROA;

**AND WHEREAS** the Corporation has agreed to a procedure in an effort to expedite grievances;

**THEREFORE**, the parties hereto have agreed to an expedited process to ensure the orderly and timely resolution of outstanding grievances.

- 1. The process will continue with the signing of this Collective Agreement and remain in effect for the life of the Collective Agreement commencing January 2020.
- 2. The parties will agree on an arbitrator, for one day only each April and October, unless mutually agreed on an alternate date, to hear predetermined cases in mediation-arbitration. The cases scheduled for Mediation-Arbitration will be agreed upon by the parties no later than the 15th of the preceding month.
- **3.** Any and all fees charged by, or costs incurred by the Mediator-Arbitrator, shall be shared equally between the Corporation and the Union.
- **4.** This arrangement for mediation-expedited arbitration is undertaken by the Union and the Corporation in an attempt, in good faith, to clear up outstanding grievances.
- 5. Any settlements reached in the mediation phase shall be without prejudice to either party, and shall not be regarded as an admission of liability by either the Corporation or the Union.
- 6. Should any dispute not be settled in the mediation phase, it may be progressed by either the Corporation or the Union to the expedited arbitration phase.

- 7. The jurisdiction of the Mediator-Arbitrator shall extend and be limited to solely the mediation and expedited arbitration of specific disputes respecting the meaning or alleged violation of any one or more provisions of a valid and subsisting collective agreement between the Union and the Corporation, or pertinent legislation, including any claims related to such provisions that an employee has been unjustly or excessively disciplined but specifically excluding any cases involving the discharge of an employee.
- 8. This special mediation-expedited arbitration process will be conducted in accordance with the instructions of the Mediator-Arbitrator, or as otherwise agreed to by the parties at the time of the commencement of the proceedings.
- **9.** Should a case not be resolved by mediation, either party may refer the matter to the Mediator-Arbitrator for final and binding resolution. The submissions of the parties shall be limited to oral presentations of a maximum duration of 20 minutes (including rebuttal) for each party, per case.
- **10.** Representations and arguments during this special process of mediationarbitration shall be restricted and limited for each case, to no more than two (2) spokespersons for the Union and two (2) spokespersons for the Corporation, per case. Legal counsel will not be permitted to attend on behalf of either party.
- **11.** The cases referred to the Mediator-Arbitrator for mediation and final and binding resolution shall be limited to those specifically agreed upon beforehand, in writing, by the Union and the Corporation. The cases will be presented to the Mediator-Arbitrator in date order: i.e.: from the oldest to the most recent.
- **12.** Each case referred to the Mediator-Arbitrator for expedited arbitration will be numbered consecutively by the Mediator-Arbitrator and he shall provide his decision orally immediately following the presentation of each case, unless otherwise agreed to between the Corporation and the Union. At the request of either of the parties, the Mediator-Arbitrator shall provide the parties with a short, written summary of his decision(s), within 30 days of the special proceedings, or as otherwise agreed to by the parties.
- **13.** The decisions of the Mediator-Arbitrator on any case referred to him for final and binding resolution under this process will not be used by the parties hereto for the purpose of establishing precedents or jurisprudence for the handling of any future cases of a like or similar nature and neither party may rely on any of theses cases to support a claim that the issue had been definitively settled by a decision on any other case. The Mediator-Arbitrator shall not, in any case, add to, subtract

from, modify, rescind, or disregard any provision of the Collective Agreement.

- **14.** Each decision of the Mediator-Arbitrator, which is made under the jurisdiction of this memorandum of agreement, shall be final and binding upon the Corporation and the Union, as well as the employee(s) involved.
- **15.** The Corporation and the Union agree that the powers of the Mediator-Arbitrator are restricted by and to these rules notwithstanding any other agreement to the contrary. The Mediator-Arbitrator shall not have the power to modify these agreed upon rules without the written consent of both the Corporation and the Union.
- **16.** In the event that the parties encounter difficulties in implementing the decision(s) of the Mediator-Arbitrator, the Corporation and the Union agree that the Mediator-Arbitrator will remain seized of each of the cases presented to him for arbitration.
- **17.** The decision of the Mediator-Arbitrator shall not be subject to appeal by either the Corporation, the Union or the employee(s) involved.
- **18.** These special mediation-arbitration sessions will be held at times and locations as mutually agreed upon by the parties.
- **19.** The foregoing shall apply to Collective Agreements No. 1 and 2.

Dave Kissack President, Unifor Council 4000 Relations

Karine Chapados Specialist Advisor, Employee VIA Rail Canada

### APPENDIX H

### LETTER ON SPLITTING OF "BEREAVEMENT LEAVE"

April 19, 1989

Mr. T. McGrath National Vice-President, Canadian Brotherhood of Railway Transport and General Workers

Dear Mr. McGrath:

During the national negotiations of the renewal of the Collective Agreements we discussed the Brotherhood's demand regarding the splitting of "Bereavement Leave".

It was the Corporation's position that the splitting of the three or five consecutive days (except rest days) was not within the intent or purpose of the Rule. The Brotherhood agreed and the demand as such was dropped. However, there was a request that some consideration be given to an employee who had taken less than the three or five consecutive days at the time of death, to be allowed the balance of the leave at a later date. The purpose of the request was to enable the employee to attend the interment when it was not part of the funeral.

The Corporation agreed that when bona fide situations of this nature arose the employees should be given appropriate consideration and that we would advise the Corporation's officers accordingly.

Yours truly,

signed A.D. Andrew) A.D. Andrew Director, Labour Relations

Amended June 8, 2016

# APPENDIX I

### LETTER ON PAYMENT FOR SICK LEAVE

Letter on Payment for Personal Leave and Sick Leave

Dear Ms. Chapados,

During the negotiations of the Memorandum of Settlement for Collective Agreement No. 1 signed on June 16, 2022, we discussed your proposal for paid personal leave and sick leave and its implications for Appendix I at some length. In resolution of your proposal, the Parties agreed to formalize the policy covering weekly rated clerical employees who are absent from duty due personal leave or sick leaves. The policy provides that such employees:

After January 1, 2023, and after three (3) months of employment, all clerical employees will become eligible for five (5) fully paid days personal leave and ten (10) fully paid days sick leave per year.

For the purposes of this agreement personal leave and sick leave will be defined and identified as per the Canadian Labour Code.

Such personal leave and sick leave encompass any personal and sick leaves identified in the Canada Labour Code and cannot be carried over from one year to the next.

Dave Kissack President Unifor National Council 4000

# APPENDIX J

#### LETTER CONCERNING THE ASSESSMENT OF DISCIPLINE

#### FOR MINOR OFFENCES WITHOUT THE NECESSITY OF HOLDING AN

#### INVESTIGATION UNDER ARTICLE 24.2

June 26, 2010

Mr. Bob Fitzgerald National Representative CAW-Canada 65 Front St. West, Room 290 Toronto Ontario M5J 1E6

Dear Mr. Fitzgerald,

During the negotiation of the Memorandum of Settlement for Collective Agreement No. 1 we discussed the Corporation's proposal to provide for the assessment of discipline for minor offences without the necessity of holding an investigation under Article 24.2 of Agreement No. 1.

In resolution of that proposal, the parties agreed to the following:

- 1. Articles 24.1 and 24.2 will be applicable to employees disciplined or discharged for a major offence.
- 2. An employee who has completed his probationary period, who is alleged to have committed a minor offence, will not be assessed discipline without having been counselled by a Supervisor in the presence of the Local Chairperson or his authorized Committee person or a designated employee.
  - (a) Upon completion of the counselling session, if deemed necessary minor discipline may be assessed immediately or within 5 days. Such discipline may range from a verbal caution, written reprimand, or up to a maximum of five (5) demerit marks and will be communicated to the employee in writing with a copy to the Local Chairperson.
  - (b) Such minor discipline assessed as a result of a minor offence if considered unjust may be appealed within 28 calendar days in accordance with the grievance procedure commencing at Step 2 of the grievance procedure in accordance with Article 24.5.

- (c) The Corporation will inform its supervisors of the intent of this approach and will monitor its use over the period of this agreement, to ensure that range of disciplinary action is utilized in a manner consistent with the nature of the offence.
- (d) Should a disciplinary action for an alleged offence be contemplated for an employee whose discipline records stands at twenty (20) demerit marks or more, an investigation must be held pursuant to Article 24

With respect to major offences, during the period of understanding all evidence will be made available to the Regional Vice-President at the Corporation office and, in advance of the hearing if he so desires. If the Regional Vice-President so wishes, the Local Chairperson may accompany him. The names and addresses of complainants or other witnesses may be withheld if considered necessary by the Corporation, unless required for subpoena by the National Vice-President of the Union.

If this reflects our understanding, would you kindly indicate your concurrence by signing below.

Yours truly,

I concur,

<u>(Signed E.J. Houlihan)</u> E.J. Houlihan Director, Labour Relations (Signed Bob Fitzgerald) Bob Fitzgerald National Representative

# APPENDIX K

# LETTER CONCERNING EMPLOYEES IN COLLECTIVE AGREEMENT NO. 2 WHO ARE MEDICALLY UNFIT TO PERFORM THEIR REGULAR

### DUTIES DUE TO PREGNANCY

May 21, 1992

Mr. T.N. Stol National Vice-President, Canadian Brotherhood of Railway, Transport and General Workers A.R. Mosher Centre 2300 Carling Avenue Ottawa, Ontario K2B 7G1

Dear Mr. Stol

This will confirm our understanding concerning employees of Collective Agreement No. 2 who are declared to be medically unfit to perform their regular duties for the duration of their pregnancy, by reason of pregnancy.

The Corporation agrees to permit these employees to claim unfilled vacancies in Collective Agreement No. 1 for which they are qualified or qualifiable within a short period of time. The Corporation agrees to consider these employees for training to claim said unfilled vacancies, provided that said training can be provided without unreasonable additional expenses to the Corporation.

The proper officer of the Corporation and the Local Chairperson of Collective Agreement No. 1 and Collective Agreement No. 2 may enter in a local agreement should an employee desire to exercise the above noted option.

The employee so permitted to claim work in the other Agreement shall be paid the appropriate rate, as specified in Collective Agreement No. 1. Their seniority while working in Collective Agreement No. 1 will be in accordance with Article 11 of that Agreement.

If the foregoing reflects our understanding, would you kindly so indicate by signing below.

Yours truly,

(signed C.C. Muggeridge) C.C. Muggeridge Department Director, Labour Relations I agree:

<u>(signed T.N. Stol)</u> T.N. Stol National Vice-President

# **APPENDIX L**

**INTENTIONALLY LEFT BLANK BY THE PARTIES** 

### **APPENDIX M**

# LETTER CONCERNING THE INTERPRETATION AND APPLICATION OF THE WORD "TERMINAL", AS USED THROUGHOUT COLLECTIVE AGREEMENT NO. 1

May 21, 1992

Mr. T.N. Stol National Vice-President, Canadian Brotherhood of Railway, Transport and General Workers A.R. Mosher Centre 2300 Carling Avenue Ottawa, Ontario K2B 7G1

Dear Mr. Stol:

During the negotiations for the renewal of Collective Agreement No. 1, you raised concerns about the interpretation and application of the word "terminal", as used throughout this agreement.

This will confirm our understanding that the Corporation and the Union, at the national level, shall meet during the life of this agreement to formulate a uniform definition of the word terminal.

In accordance with the above, the Corporation shall canvass its regional officers to determine the present interpretation and applications of the word terminal. This information shall be compiled and forwarded to the National Vice-President of the Union prior to the aforementioned meeting.

If the foregoing accurately reflects our mutual understanding, would you kindly indicate your concurrence by signing below.

Yours truly,

(signed C.C. Muggeridge) C.C. Muggeridge Department Director, Labour Relations

I agree:

(signed T.N. Stol) T.N. Stol National Vice-President CBRT & GW

# **APPENDIX N**

**INTENTIONALLY LEFT BLANK BY THE PARTIES** 

# APPENDIX O

# LETTER CONCERNING THE BREAK PERIODS FOR THE EMPLOYEES WORKING IN THE TELEPHONE SALES OFFICE

June 14, 2001

Mr. Tom Wood National Representative

Dear Sir:

During the current round of negotiations for renewal of Collective Agreement No. 1, the Corporation and the Union discussed the issue of break periods for their members working exclusively with display terminals in the Telephone Sales Offices. To address the concerns expressed by the Union, those employees will be entitled to the following break periods:

- 1. A thirty (30) minute lunch period.
- 2. A fifteen (15) minute break period within the first four hours of the shift, as assigned by the Corporation.
- 3. A further fifteen (15) minute break period within the last four hours of the shift, as assigned by the Corporation.
- 4. An additional twenty (20) minutes for breaks which may be taken at the discretion of the employee, in more than one segment, subject to the following conditions:
  - a) it may not interfere with operational requirements;
  - b) it may not be added to the lunch break period or either of the fifteen (15) minute breaks;
  - c) it may not be used for late arrival or early departure;
  - d) it is non-cumulative and failure to use it fully will not result in overtime.

# FOR THE CORPORATION:

(signed by B.E. Woods) B.E. Woods Director, Labour Relations

# FOR THE CORPORATION:

(signed by Dino Trubiano) Dino Trubiano Senior Officer, Labour Relations

I CONCUR:

(signed by Doug Olshewski) Doug Olshewski National Representative CAW/TCA Canada

## APPENDIX P

#### LETTER CONCERNING THE APPLICATION

#### OF ARTICLE 4.26

201 Ash Avenue Montreal, Quebec H3K 3K2 201, avenue Ash Montréal (Québec) H3K 3K2

Labour Relations Operations Relations du Travail Exploitation

June 29, 2001

Mr. Gary Fane Director, Transportation CAW/TCA Canada

Dear Sir:

With regards to the establishment of a national ratio for part-time employees, this will confirm, concerning the application of clause 4.26, that it is not the intention of the Corporation to have regular full-time employees work part-time hours on a "Maintenance of Earnings" basis as stipulated in clause 4.26

Yours truly,

(signed by Bannon E. Woods) Bannon E. Woods Director, Labour Relations

# **APPENDIX Q**

## LETTER OF UNDERSTANDING

### LETTER CONCERNING THE APPLICATION OF ARTICLE 11.9

July 19, 2007

Mr. D. Olshewski National Representative CAW-Canada 1376 Grant Ave. Winnipeg, Manitoba R3M 3Y4

Dear Mr. Olshewski,

SUBJECT: Application of Article 11.9 of Collective Agreement No. 1 and Article 11.6 of Collective Agreement No. 2

The issue was raised in negotiations regarding the manner in which we administer employees promoted to excepted positions.

The problems arise when promoted employees remain staffed on unionized positions while performing management duties. As their status is not changed in VIP, there is no trigger to stop the payment of union dues or to start the 12-month clock to freeze seniority.

To resolve the matter, the Corporation has agreed to ensure that employees who are promoted are staffed accordingly. A copy of the staff form should be forwarded to the Regional Representative. Also, the designated union officer shall be provided with a quarterly report by e-mail of employees who are promoted to a permanent or temporary position with the information available by region, province and by collective agreement.

Thank you for your cooperation in this matter.

Yours truly,

(signed by E.J. Houlihan) Edward J. Houlihan Director, Labour Relations

# **APPENDIX R**

# SKILL SETS

Job Title	Group	Skill Set No.	Comments
Classified Labourer	Non Clerical	1	
Janitor	Non Clerical	1	
Lead Hand Classified Labourer	Non Clerical	1	
Office Services Clerk	Clerical	1	
Red Cap	Non Clerical	1	
Red Cap Captain	Non Clerical	1	
Truck Driver/5-ton Pettibone Operator	Non Clerical	1	With license
Forklift Operator / Tow Motor	Non Clerical	2	
Senior Station Services Agent	Non Clerical	2	
Station Services Agent	Non Clerical	2	
Ticket Seller / Examiner	Non Clerical	2	
Shipper – Receiver	Clerical	3	
Stock Attendant	Non Clerical	3	
Stock Checker	Non Clerical	3	
Stock Keeper	Non Clerical	3	
Stores Attendant	Non Clerical	3	
Senior StoresAttendant	Non Clerical	3	
Locomotive Attendant	Non Clerical	4	
Communications Operator	Clerical	5	
Administration Clerk	Clerical	6	
Administration Support Clerk	Clerical	6	
Data Entry Clerk	Clerical	6	
Word Processor Operator	Clerical	6	
Work Order Copy Clerk	Clerical	6	
Word Processor Specialist	Clerical	6	
General Operations Clerk	Clerical	7	
Inventory Control Clerk	Clerical	7	
Senior ESCAttendant	Clerical	7	
Senior Administration Clerk	Clerical	7	

# APPENDIX R SKILL SETS

Job Title	Group	Skill Set No.	Comments
Accounting Clerk	Clerical	8	
Payroll Clerk	Clerical	8	
Senior Accounting Clerk	Clerical	8	
Senior Payroll Clerk	Clerical	8	
		-	
Control Clerk (#1 admin.)	Clerical	9	
Employee Services Clerk	Clerical	9	
Pension Admin. Clerk	Clerical	9	
Control Clerk (#2 admin.)	Clerical	10	
Senior Control Clerk (#2 admin.)	Clerical	10	
	Ciencal	10	
Senior Computer Operator	Clerical	11	
Counter Sales Agent	VIANET	12	
Internet Agent	VIANET	12	Testing of writing skills
Rate & Refund Agent	VIANET	12	
Special Traffic Agent	VIANET	12	Testing of writing skills
Telephone Sales Agent	VIANET	12	
VIANET Text Clerk	Clerical	12	
		10	
Senior Counter Sales Agent	VIANET	13	
Senior Telephone Sales Agent	VIANET	13	
Senior Trade Desk Agent	VIANET	13	
Senior Internet Agent	VIANET	14	
In-Charge Ticket Office	Clerical	15	
		40	
In-Charge Station	Clerical	16	
In-Charge ESC	Clerical	17	
Payroll Clerk (Running Trades)	Clerical	18	
Senior Payroll Clerk (Running Trades)	Clerical	18	
Senior Administration Clerk (Systems Support)	Clerical	19	

Any position that is created, changed or modified will be placed in the appropriate skill set, as agreed between the Corporation and the Union in accordance with the Collective Agreement.

# **APPENDIX S - REMOVED**

### INTENTIONALLY LEFT BLANK BY THE PARTIES

# APPENDIX T

### LETTER CONCERNING RECOGNITION PROGRAM

June 26, 2010

Mr. Bob Fitzgerald National Representative CAW-Canada 65 Front St. West, Room 290 Toronto Ontario M5J 1E6

Dear Mr. Fitzgerald,

The Corporation and the Union recognize VIA's future is dependent on continuous growth in ridership and revenue.

The Union is recognized as the sole bargaining agent with respect to working conditions in accordance with Article 2.1 of Collective Agreement.

Prior to implementation of any performance recognition programs designed to grow ridership or revenue the Corporation will consult with the designated Union Representative regarding the program to ensure that they are in keeping with the recognition and rights of the Union.

Recognition programs will be the subject of a mutual agreement between the proper officer of the Corporation and the designated National or Regional Representative of the Union.

Yours truly,

I concur,

<u>(Signed E.J. Houlihan)</u> E.J. Houlihan Director, Labour Relations (Signed Bob Fitzgerald Bob Fitzgerald National Representative

### APPENDIX U

#### LETTER CONCERNING MATERNITY-RELATED REASSIGNED AND LEAVE

June 26, 2010

Mr. Bob Fitzgerald National Representative CAW-Canada 65 Front St. West, Room 290 Toronto Ontario M5J 1E6

Dear Mr. Fitzgerald,

- (1) An employee who is pregnant or nursing may, during the period from the beginning of the pregnancy to the end of the twenty-fourth week following the birth, request the employer to modify her job functions or reassign her to another job if, by reason of the pregnancy or nursing, continuing any of her current job functions may pose a risk to her health or to that of the fetus or child.
- (2) An employee's request under (1) must be accompanied by a certificate of a qualified medical practitioner of the employee's choice indicating the expected duration of the potential risk and the activities or conditions to avoid in order to eliminate the risk.
- (3) An employer to whom a request has been made under (1) shall examine the request in consultation with the employee and, where reasonably practicable, shall modify the employee's job functions or reassign her.
- (4) An employee who has made a request under (1) is entitled to continue in her current job while the employer examines her request, but, if the risk posed by continuing any of her job functions so requires, she is entitled to and shall be granted a leave of absence with pay at her regular rate of wages until the employer.

- (a) modifies her job functions or reassigns her, or
- (b) informs her in writing that it is not reasonably practicable to modify her job functions or reassign her, and that pay shall for all purposes be deemed to be wages.

Yours truly,

I concur,

(Signed E.J. Houlihan) E.J. Houlihan Director, Labour Relations (Signed Bob Fitzgerald) Bob Fitzgerald National Representative

## APPENDIX V

Articles 4.1 and 4.11 Collective Agreement #1

As a result of the bargaining for the renewal of the Collective Agreement, the parties agreed to a temporary amendment to Articles 4.1 and 4.11 for the duration of the agreement only.

Article 4.1 Thirty days after ratification of the agreement, eight consecutive hours of service inclusive of the meal period shall constitute a day's work.

Article 4.11

Thirty days after the ratification of the agreement, notwithstanding the provisions of Articles 4.5 and 6, regular assignments consisting of four days of 10 hours may be established at the discretion of the Corporation.

Dated at Gatineau the 8th day of June 2016

### For the Corporation

For the Union

Ed Houlihan Director, Employee Relations Bob Fitzgerald National Representative

### **APPENDIX W**

#### **RED-CIRCLED FULL-TIME EMPLOYEES**

In the event a full-time employee holding a permanent position is facing lay-off they will have the option to accept lay-off status or displace a junior part-time employee within their J-SET. In these circumstances only and notwithstanding the terms of the Supplemental Agreement, the full-time employee will be "Red- Circled" in that position and will be allowed to return to a full-time permanent position when their seniority permits.

If the employee chooses to accept part-time work, they are obligated to bid on all full-time vacant positions for which they are qualified in their J-SET when such is bulletined. Should the employee fail to bid on a permanent full-time job, they will become a part-time employee and will lose their red-circled status. Upon recall, the junior laid off employee may be forced to any unfilled full-time vacancy. The red-circled employee may also choose to bid to any full-time permanent vacancies on their Region. The provisions of Articles 12.15 and 13 will apply.

Once the red-circled full-time permanent employee is awarded a full-time permanent position, their special status is at an end.

The agreement will commence January 1, 2016 and will expire December 31, 2019.

Yours Truly,

I concur,

(Signed E.J. Houlihan) E.J. Houlihan Director, Labour Relations (Signed Bob Fitzgerald) Bob Fitzgerald National Representative

# APPENDIX X

### **CANADIAN ARMED FORCES RESERVES AT VIA RAIL**

Reservists are members of the Canadian Armed Forces who serve on a part-time basis and hold civilian jobs. Members of the Reserve Force are required to undergo training of up to 15 days annually. They are entitled to an unpaid leave of absence for this period under Article 247.5(1) of the Canada Labour Code.

To support the reservists working at VIA Rail and attract members of the Reserve Force to join VIA Rail, we wish to enhance the minimum standards prescribed under the Code for our employees and your members.

The Corporation proposes the following;

- Leave for military training payment of top up benefit: 100% of base salary (the difference between the employee's gross salary and the pay received by the Reserve Force) -1 leave per year, a maximum of 15 working days per year;
- Group Insurance all coverages maintained (Note: no benefits paid if expenses, claims or disability arise from war or insurrection.)
- Pension Plan the employee may choose to maintain participation (they must pay his/her contributions)

The Corporation wishes to implement these changes as soon as possible and Council 4000 recognizes the importance of these improvements and agrees to make these changes to the leave provisions for members of the Reserve Force. The parties agree that these provisions will be incorporated into Collective Agreements 1 and 2 upon renewal.

Agreed April 14, 2016

<u>(Signed E.J. Houlihan)</u> E.J. Houlihan Director, Labour Relations (Signed Bob Fitzgerald) Bob Fitzgerald National Representative

# **APPENDIX Y**

#### LETTER CONCERNING CONSENT FOR UNREDUCED PENSIONS

December 22, 2020

Mr. Dave Kissack, President National Council 400 UNIFOR

Dear Mr. Kissack:

This refers to the recently concluded agreement covering Off-train employees (Collective Agreement No. 1).

Although the question of pensions does not form part of any collective agreement, this confirm that we will continue granting consent to an unreduced pension, for the life of the current collective agreements commencing on January 1st, 2020.

The Corporation will continue the practice of granting consent to an unreduced pension (55/85 rule) for the life of the current collective agreement commencing January 1st, 2020.

Yours truly,

Karine Chapados Specialist advisor, employee relations

# APPENDIX Z

BUNDLE: ALL LEAVES PROVIDED FOR IN THE CANADIAN LABOUR CODE

# PERSONAL LEAVE

Provided an employee has three (3) consecutive months of continuous employment, they will be granted five (5) days of personal leave, including three (3) paid leave days and two (2) unpaid leave days per calendar year. An employee has not completed three months of continuous employment, they will still be entitled to five days of leave without pay. Employees are entitled this leave as follows:

- •treat an injury or illness;
- •take care of health obligations for any member of their family or care for them;
- •take care of obligations related to the education of any family member under age 18;
- •manage any urgent situation that concerns you or a family member;
- •attend your citizenship ceremony under the Citizenship Act, or;
- •manage any other situation prescribed by regulation.

Leave may be taken in one (1) or more periods, however the Corporation may require that each period be at least one (1) day. Employees may be requested to provide supporting documents concerning the reasons for the leave.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

## Appendix AA

## Letter concerning composite position

The Corporation will advise the Union leadership of the creation of any future composite position.

## Appendix AB

#### VIA RAIL INC. (Herein referred to as the "Corporation")

AND

## UNIFOR, COUNCIL 4000

(Herein referred to as the "Union")

## SUBJECT: Vacation Entitlement Calculations – Application starting January 2023

- **WHEREAS** During the 2021-2022 bargaining sessions, the parties have identified that the calculation for employees' vacations, or moreover, the triggering point which transitions the employee between vacation entitlement categories under Articles 9.1 through 9.5 of Collective Agreements One and Two, has been applied by utilizing calculations based on Cumulative Compensated Service rather than by Continuous Employment Relationship from service date.
- **WHEREAS** The parties recognize for the sole purpose of vacation calculation, twelve (12) calendar months constitutes one years' service.
- **WHEREAS** The parties recognize that the standard indicator for movement between vacation entitlement categories is based on the anniversary of the employee's date of hire (service date).
- WHEREAS The parties recognize that the term Continuous Employment Relationship encompasses active service, periods of lay off for determination of vacation allotment, periods of absence as defined under Article 9.11, and/or any other circumstances as mutually agreed between the parties.

## THEREFORE, the parties agree to the following:

- 1. Going forward the triggering point which transitions employees between entitlement categories under Articles 9.1 through 9.5 for vacation calculation purposes will be based on annual increments from the employee's date of hire (service date) and will apply to the following calendar year.
- 2. Actual days of vacation entitlement days under Articles 9.1 through 9.5 of Collective Agreements One and Two will be maintained as based on the days worked or the major portion thereof in the previous calendar year.

- 3. The current practice for calculating what constitutes a "day worked" for vacation calculation purposes when applied to spareboard employees in Agreement Two will remain in effect whereas eight hours constitutes one day of work.
- 4. This Agreement is made with precedent serves to rescind any previous practices which refer to Cumulative Compensated Service for the calculation of vacation entitlements. The use of Cumulative Compensated Service will remain in effect as defined under Article 1.8 of Collective Agreements One and 1.1(u) of Collective Agreement Two (whereas 250 days cumulative compensated service constitutes one years' service) for the purposes of calculating Pensionable Service and any protections under the Supplemental Agreements (ESIMA).

## COMPASSIONATE CARE LEAVE

An employee, can take up to twenty-eight (28) weeks of unpaid compassionate care leave within a fifty-two (52) week period to look after a family member as defined in article 29.1 in Collective Agreement 1 and article 27.2 in Collective Agreement 2, who has a serious medical condition with a significant risk of death.

The leave begins during one of the following weeks, whichever occurs first:

- the week the health care practitioner signs the medical certificate;
- the week the health care practitioner examines the gravely ill family member, or;
- the week the family member becomes gravely ill, if the health care practitioner can determine that date (for example, the date of the test results).

The leave ends when:

- the twenty-eight (28) weeks of compassionate care are complete;
- the gravely ill family member dies or no longer requires care or support, or;
- the fifty-two (52) week period expires.

Two (2) or more employees can share compassionate care leave when looking after the same family member. However, the total amount of leave taken by all cannot be more than twenty-eight (28) weeks within the fifty-two (52) week period.

An employee can consecutively take the leave related to critical illness to care for the same person, if eligible. However, they cannot take compassionate care leave if one (1) or more employees are taking the leave related to critical illness in respect of the same person.

The employee must give the Corporation written notice, as soon as possible, advising of the reason(s) for the leave and the intended length of the leave. If the leave taken is more than four (4) weeks and the employee wants to change the length of the leave, they must provide the Corporation with four (4) weeks' notice, unless there is a valid reason why this is not possible.

If requested in writing by The Corporation, the employee must provide a certificate from a health care practitioner within fifteen (15) days of their return to work. This certificate must state that the family member has a serious medical condition and as a result, there was significant risk of death within twenty-six (26) weeks.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee relations VIA Rail Canada

## CRITICAL ILLNESS LEAVE

An employee who is a family member as defined in article 29.1 in Collective Agreement 1 and article 27.2 in Collective Agreement 2, of a critically ill child or adult, is eligible to take:

- up to thirty-seven (37) weeks of unpaid leave in a fifty-two (52) week period to provide care or support to the child under 18 years of age, and;
- up to seventeen (17) weeks of unpaid leave in a fifty-two (52) week period to provide care or support to the adult.

The leave begins during one of the following weeks, whichever occurs first:

- on the first day of the week the medical certificate is issued or;
- the day from which the health care practitioner certifies that the child or adult is critically ill.

If the child or adult dies while the employee is on leave, the leave ends on the last day of that week.

If two (2) or more children are critically ill, the employee is eligible for separate leaves of thirtyseven (37) weeks with respect to each affected child.

The employee must give the Corporation written notice, as soon as possible, advising of the reason(s) for the leave and the intended length of the leave. If the leave taken is more than four (4) weeks and the employee wants to change the length of the leave, they must provide the Corporation with four (4) weeks' notice, unless there is a valid reason why this is not possible.

The employee must also provide the Corporation with a certificate from a health care practitioner. The certificate must state that the child or adult is critically ill or injured and requires the care or support of one (1) or more of their family members.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

## DEATH OR DISAPPEARANCE LEAVE

An employee whose child is under 18 years of age and has disappeared or died as a result of a probable crime under the Criminal Code, is eligible to take up to:

- fifty-two (52) weeks of unpaid leave in the case of a missing child, starting on the day the disappearance occurs, and;
- 104 weeks of unpaid leave if the child has died, starting on the day the death occurs

To take this leave, the employee must be the:

legal parent

adoptive parent

a person with whom the child was placed for the purpose of adoption, or

an individual with legal custody or guardianship of the child who has died or disappeared

An employee is not eligible for this leave if they are charged with the crime, or, if it is probable, considering the circumstances, that the child was a party to the crime.

To obtain this leave, the employee must advise the Corporation in writing as soon as possible of the reason(s) for the leave and the planned length of the leave. They must notify the Corporation in writing of any changes in the length of the leave as soon as possible.

If the length of the leave is longer than four (4) weeks, the employee must give the Corporation at least four (4) weeks' notice of any change in length of leave, unless there is a valid reason why this is not possible. If the leave is for a disappeared child who is then found dead, employee can change the type of leave by notifying the Corporation in writing as soon as possible.

The Corporation may request the employee to provide a proof of entitlement to the leave, such as a police report.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

## LEAVE FOR ABORIGINAL PRACTICES

Provided an Aboriginal employee (Indian, Inuit or Métis) has three (3) consecutive months of continuous employment, they will be granted five (5) unpaid leave days per calendar year to engage in traditional Aboriginal practices including:

- fishing
- hunting
- harvesting
- traditional ceremonies
- all practices prescribed by regulation

The leave may be taken over more than one (1) period, however the Corporation may require that each period be at least one (1) day. Employees may be requested to provide supporting documents concerning the reasons for the leave

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

## MATERNITY LEAVE

An employee who becomes pregnant shall be entitled to seventeen (17) weeks of maternity leave. They can take this leave any time during the period that:

- begins thirteen (13) weeks before the expected date of birth, and;
- ends seventeen (17) weeks after the actual birth date

The employee must provide their supervisor with a certificate from a health care practitioner confirming that they are pregnant. Written notice at least four (4) weeks before starting the leave is required. This notice must advise of the length of the leave.

If their child was not born during the seventeen (17) weeks of your maternity leave, the maternity leave is extended until the date of the birth.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

## PARENTAL LEAVE

A natural or adoptive parent is eligible for up to sixty-three (63) weeks of unpaid parental leave. Parents, who both work for the Corporation, may share parental leave in order to access an additional eight (8) weeks of leave. Parents who share the parental leave have access to seventy-one (71) weeks of unpaid parental leave. The employee can take this leave any time during the following:

- within the seventy-eight (78) week period starting the day their child is born, or;
- the day their child comes into their care

The employee must also give the Corporation written notice at least four (4) weeks before starting their leave. This notice must advise the Corporation of the length of the leave.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

## LEAVE FOR VICTIMS OF FAMILY VIOLENCE

Provided an employee had a continuous employment relationship of three (3) consecutive months with the corporation, they will be granted a maximum of ten (10) days, including five (5) paid leave days and five (5) unpaid leave days per calendar year, to engage in activities related to family violence for themselves or for their child, such as:

- to seek medical attention for themself or their child in respect of a physical or psychological injury or disability;
- to obtain services from an organization which provides services to victims of family violence;
- to obtain psychological or other professional counselling;
- to relocate temporarily or permanently;
- to seek legal or law enforcement assistance or to prepare for or participate in any civil or criminal legal proceeding, or;
- to take any measure prescribed by regulation.

The leave may be taken over more than one (1) period, however the Corporation may require that each period be at least one (1) day. Employees may be requested to provide supporting documents concerning the reasons for the leave. If the employee has not completed three months of continuous employment, they will still be entitled to ten days of unpaid leave.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

## LANGUAGE TRAINING

June 26, 2010

Mr. Bob Fitzgerald National Representative CAW-Canada 65 Front St. West, Room 290 Toronto Ontario M5J 1E6

Dear Mr. Fitzgerald,

Further to our discussions during this round of bargaining the Corporation has agreed to provide the opportunity for two suitable full-time employees to receive French Language training in 2011.

One (1) will be provided in Toronto and one (1) will be provided in the Western region.

Yours truly,

(Signed Edward J. Houlihan) Edward J. Houlihan Director, Labour Relations

#### VACATIONS SOUTHWESTERN ONTARIO

June 26, 2010

Mr. Bob Fitzgerald National Representative CAW-Canada 65 Front St. West, Room 290 Toronto Ontario M5J 1E6

Dear Mr. Fitzgerald,

During the negotiation of Collective Agreement No. 1, the Union raised the issue of employees in Southwestern Ontario requesting vacation outside the summer season.

It is agreed between the parties that consent to such a request not be unreasonably withheld.

Overtime costs to be incurred by the Corporation are a valid reason to withhold consent.

If the foregoing clearly reflects our understanding, would you please so indicate by signing below.

Yours truly,

I concur,

(Signed E.J. Houlihan) E.J. Houlihan Director, Labour Relations (Signed Bob Fitzgerald) Bob Fitzgerald National Representative

## LETTER ON SHIFT TRADES

June 26, 2010

Mr. Bob Fitzgerald National Representative CAW-Canada 65 Front St. West, Room 290 Toronto Ontario M5J 1E6

Dear Mr. Fitzgerald,

During the negotiation of Collective Agreement No. 1, the Corporation raised the issue of the Shift trade procedure being national in scope.

This will confirm our understanding that the Union has opted for a six-month trial period for this procedure without prejudice. The union may opt to withdraw from the procedure after the 6-month period.

If the foregoing clearly reflects our understanding, would you please so indicate by signing below.

Yours truly,

I concur,

<u>(Signed E.J. Houlihan)</u> E.J. Houlihan Director, Labour Relations (Signed Bob Fitzgerald) Bob Fitzgerald National Representative

## RATE OF PAY OF MONTREAL STATION IN-CHARGE

June 12, 2013

Mr. Bob Fitzgerald National Representative CAW-Canada 205 Placer Court Toronto, Ontario M2H 3H9

Dear Mr. Fitzgerald,

During the negotiations of the Memorandum of Settlement for Collective Agreement No. 1 we discussed the Union's proposal to modify the rate of pay for the Montreal Central Station In-charge position.

To address the issue, the parties have agreed as follows:

Effective August 1, 2013, in recognition of the particular duties and responsibilities of this position, the Montreal Central Station In-charge will be compensated at the premium rate of \$50.00 per week above that of the senior counter sales agent rate.

Signed at Montreal the \_\_\_\_\_day of June 2013.

<u>(Signed E.J. Houlihan)</u> E.J. Houlihan Director, Employee Relations (Signed Bob Fitzgerald) Bob Fitzgerald National Representative

#### **JOB SHARING**

#### Purpose

The Union and the Corporation jointly developed a job-sharing arrangement to allow employees to temporarily share a position or assignment that better meets their particular needs without changing their permanent status with the Corporation.

#### 1. Eligibility

To be eligible to enter into a job-sharing arrangement, both the "Host" person whose job will be shared and the "Guest", the one sharing the position, must hold full time assignments. The Guest must be an active employee and qualified on the job to be shared. Neither Host nor Guest can be subject to a lock-in as a result of training or subject to the obligations of Maintenance of Basic Rates (MBR), although they can waive their rights to MBR if they wish. The Host and/or Guest will be advised in writing of the consequences of waiving their rights.

2. Duration of Job-Sharing Arrangement

A job-sharing arrangement will be for six calendar months. The Host and/or Guest may terminate the arrangement earlier upon 30 calendar days' notice in writing. At the end of the six-month period or until the next general bid, the Host and Guest may apply for an extension of the job-sharing arrangement. If they are selected, they will have a further 6 calendar months or until next general bid. There can be no further extensions of the job-sharing arrangement after one year to the same Host and Guest unless the maximum amount of job-sharing positions allowable in that region has not been maintained.

#### 3. Application Process

For Collective Agreement, No. 1 and No. 2, a total of 2 job sharing opportunities will be distributed as follow, for a total of 8 opportunities:

- 2 opportunities for the Eastern region (Halifax Moncton)
- 2 opportunities for Corridor East
- 2 opportunities for Central region
- 2 opportunities for the Western region

\*This may be readjusted based on operational needs.

If any job-sharing opportunities are not used in a region, the number of opportunities may be increased by a corresponding number in other regions if they have additional demand. If accepted for a job-sharing arrangement, the Guest's permanent position will be posted for bid in accordance with Article 12.1 (long term temporary position or 12.6) under Agreement #1 and Article 12.3 under Agreement #2.

4. Selection process

a) Individual employees at their work location will be responsible for finding their own partners to job share.

b) Applications received will be considered, based on seniority, seniority of the host governing.

- c) Applications will be considered by date of application.
- 5. Review of Job Hours

In the event there is a job-sharing arrangement for a position under Agreement #1, the Corporation may offer the Host and the Guest a new distribution of work hours as long as it better meets operational needs and agreed by both the Host and the Guest.

The change of hours agreed will not result in changes to current schedules or positions. If agreed, the hours of work would be changed for the duration of the job-sharing arrangement only, on a without prejudice or precedent basis. The hours of work for the Host position would revert to the original hours of work at the end of the job-sharing arrangement.

- 6. Rules for Job Sharing
- a) The hours of the assignment or position will be shared as equally as possible with no more than one week between days of work for Agreement #1 and Agreement #2 Corridor, two weeks for Agreement #2, Long hauls and Remotes.
- b) Both the Host and Guest must maintain an attendance rate of not less than 95% in any continuous 3-month period. Workers Compensation and STD will not be included in the 95% rate.
- c) Guest and Host will continue the benefit coverage they enjoyed prior to job sharing, either single or family coverage, for the duration of the job-sharing arrangement.
- d) Guest nor Host may take promotional training or other training requiring a lockin while in a job-sharing arrangement.

- e) The employees will be permitted to declare for work (including "Standby") on their layover period, however they will only be paid overtime after an accumulation of 160 hours in a four (4) week period Corridor (320 long haul). (The work accepted must be able to be completed during the layover period).
- f) If the Guest is displaced from their permanent position upon completion of the jobsharing agreement, they can exercise seniority in accordance with the Collective Agreement.
- g) If the Host or Guest position is abolished while in a job-sharing agreement, the agreement will be terminated on the date of abolishment.
- i) For Agreement #2 employees, the guarantee payment available for the assignment will be divided as equally as possible between the Host and the Guest.
- **j)** The possibility of Host having a "Job Share" covered by the spareboard may be authorized under special circumstances established by the corporation in the event a Guest is not available.
- 7. Cancellation of Job-Sharing Arrangement

The Guest, Host, Union or Corporation may cancel the any specific job-sharing arrangement at any time upon 30 days' written notice.

However, if any of the following events occur, the job-sharing arrangement will may be cancelled automatically with 30 days to return to their original assignments:

- a) Failure of either Guest or Host to maintain and attendance rate of 95% or greater in any continuous 3-month period.
- b) The Host is displaced from his/her permanent position
- c) If either Guest or Host are absent or will be absent due to illness for more than 30 calendar days
- d) If either the Guest or Host are injured and in receipt of Worker's Compensation Benefits for 15 calendar days or more.
- 8. Duration

The job-sharing agreement will commence January 1, 2020. The agreement will be in force for the life of the collective agreement.

## 9. Statutory Holiday Pay

If both the Host and the Guest are on the scheduled layover days of the assignment during a statutory holiday, they will both be paid 4 hours holiday pay. If one of the two (2) employees works on a statutory holiday, only that employee will receive the corresponding eight (8) hour holiday pay.

Dave Kissack President, Unifor Council 4000

ado5

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

#### Article 38.6 of the Collective agreement

December 2nd, 2020,

Ms. Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada, Inc. 3 Place Ville-Marie, Suite 500 Montréal, Québec H3B 2C9

Ms. Chapados:

During the 2019-2020 contract renewal negotiations, the Union and the Corporation discussed the need for a common interpretation of Article 38.6 of Collective Agreement No. 1. The Union expressed concern that our members were not being provided with outerwear in a timely fashion in certain locations across the country.

The Corporation committed, when an employee whose position regularly requires them to work outside during the colder periods of the year or is required to work in colder environments inside the Corporation's facilities (for example, a refrigerator or freezer), to ensuring that individual employees are issued with suitable outerwear within the guideline of the uniform policy or upon request of the employee to their manager. The manager will furthermore take positive steps to ensure that each employee possesses the appropriate Corporation-supplied apparel within a reasonable period before the beginning of the colder season.

If the foregoing clearly reflects our understanding, please so indicate by signing below.

Sincerely,

Dave Kissack President, Unifor Council 4000

#### LETTER OF UNDERSTANDING

#### BETWEEN

## VIA RAIL CANADA INC (hereby called "The Corporation")

#### AND

#### UNIFOR, National Council 4000 (hereby called "The Union")

#### SUBJECT: Collective's agreements send by email.

**WHEREAS** the collective agreement which is currently being negotiated is of short duration.

The Parties Agree to The Following

 For the during of this collective agreement, the parties agree to amend Article 27.14 of Collective Agreement 1 and 28.12 of Collective Agreement 2 to be read as follow:

"The parties agree once language verification and translation have been completed, the Corporation will immediately, no later than 90 days, provide to the President of Council 4000 and regional representatives and all existing and future employees with an updated electronic version of Collective Agreements No.1 and No. 2, as well as all Letters of Understanding, Health and Safety Agreement, Benefit and LTD Plans, ESIMA, and any other related documents by email. An employee can request a copy of the collective agreement on an electronic memory stick by sending an email to AskRH."

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

#### LETTER OF UNDERSTANDING

## BETWEEN

## VIA RAIL CANADA INC (hereby called "The Corporation")

## AND

### UNIFOR, National Council 4000 (hereby called "The Union")

#### SUBJECT: Ongoing Labour Relations Committee

- **WHEREAS** the collective agreement which is currently being negotiated is of short duration.
- **WHEREAS** in order to maintain the dialogue between the Corporation and the Union and to work collaboratively to discuss the terms of the future collective agreement and issues arising throughout the term of the current collective agreement.
- **WHEREAS** in the years to come, The Corporation will be subject to many new initiatives that and it will serve in the best interests of both parties to hold open and frank discussions.

The Parties Agree to The Following

Within thirty (30) days of the ratification of this Collective Agreement, an Ongoing Labour Relations Committee will be created, consisting of four (4) representatives from VIA Rail and four (4) representatives from Unifor, Local 4000. The representatives will be chosen according to the topics discussed

1. The mandate of this ongoing Labour Relations Committee will be to discuss topics and their potential impact to the bargaining units and collective agreements such as the arrival of the new fleet and any other ongoing issues between the two parties.

- 2. The ongoing Labour Relations Committee will meet once monthly. When necessary additional dates may be scheduled.
- 3. The Corporation, and the Union will send a list of topics to be discussed one week before the meeting is held

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

### EDUCATIONAL LEAVE

## LETTER OF UNDERSTANDING

#### BETWEEN

## VIA RAIL CANADA INC (hereby called "The Corporation")

#### AND

## UNIFOR, National Council 4000 (hereby called "The Union")

#### SUBJECT: Educational Leave (collective agreement 1)

- **WHEREAS** the collective agreement which is currently being negotiated is of short duration.
- WHEREAS in order to maintain the dialogue between the Corporation and the Union and to work collaboratively to discuss the terms of the future collective agreement and issues arising throughout the term of the current collective agreement.
- **WHEREAS** in the years to come, The Corporation will be subject to many new initiatives that and it will serve in the best interests of both parties to hold open and frank discussions.

#### The Parties Agree to The Following:

- 1) Based on operational requirements, leave of absence for educational purposes may be granted to employees after they have completed 12 months of cumulative compensated service (CCS).
- Employees requesting educational leave will not be entitled to take vacation during the calendar year preceding the leave and must submit their request for leave prior to the closing period for the vacation bids.

- 3) Vacation pay will be paid out at the commencement of the leave period.
- 4) Employees granted an educational leave must commit to being available for relief work during school breaks. Failure to fulfill the obligation to relief work will result in the leave being cancelled.
- 5) Employees who return to the service between school terms, or prior to terminating the educational course for which leave of absence has been granted, will not be permitted to exercise their seniority to displace and will be required to bid for an open position which they are qualified for.
- 6) The designated National or Regional Representative of the Union will be informed when such leaves are granted.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

#### LETTER OF UNDERSTANDING

#### BETWEEN

## VIA RAIL CANADA INC (hereby called "The Corporation")

#### AND

## UNIFOR, LOCAL 4000 (hereby called "The Union")

#### SUBJECT: Training / Difficult Customer Interactions

- **WHEREAS** the collective agreement which is currently being negotiated is of short duration;
- WHEREAS the current negotiation, the Union has expressed the need for all VCC employees to benefit from additional training on how to manage difficult customer interactions
- **WHEREAS** The Corporation understands the importance of offering training adapted to the realities of VCC.

#### THE PARTIES AGREE TO THE FOLLOWING:

1. The Corporation is committed to reviewing existing processes and where required, modify associated agent training and tools including job aids. As deemed necessary by the Corporation, subsequent training to all VCC employees in 2021.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

#### LETTER OF UNDERSTANDING

#### BETWEEN

## VIA RAIL CANADA INC (hereby called "The Corporation")

#### AND

#### UNIFOR, LOCAL 4000 (hereby called "The Union")

# SUBJECT: Wage Scale - Reclassify Senior Telephone Agent in Moncton and Montreal

- **WHEREAS** the collective agreement which is currently being negotiated is of short duration;
- **WHEREAS** the current negotiation, the Union has expressed the need for a job evaluation for the Senior Telephone Agent.

#### THE PARTIES AGREE TO THE FOLLOWING:

**1.** The Corporation is committed to bring this position to the job evaluation committee to determine if there should be an increase for this position.

Dave Kissack President, Unifor Council 4000

Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada

Montreal, December 3rd, 2020

Mr. Dave Kissack, President National Council 400 UNIFOR

## Women's Advocacy Program at VIA Rail for Council 4000

#### Women's Advocacy Program at VIA Rail

This refers to our recent discussions at the bargaining table pertaining to unique situations faced by women in the workplace.

The parties recognize that female employees may sometimes need to discuss with another woman matters such as violence or abuse at home or workplace harassment. They may also need to find out about specialized resources in the community such as counselors or women's shelters to assist them in dealing with these and other issues.

Therefore, in addition to the current resources such as the Corporation-sponsored Employee Assistance Program, the Corporation and Union agree to implement the Women's Advocacy Program at VIA Rail subject to the following terms and conditions:

- The Corporation agrees to a maximum of five (5) Women's Advocates positions, one per region. Consideration for the foregoing will include at least one (1) French-speaking advocate for the province of Quebec to be included in the total number of regional Women's Advocate's positions.
- 2. The Women's Advocates will be jointly selected by the Corporation and the Union from a group of female employees nominated by the Union.
- 3. The successful candidate(s) will agree to perform the Women's Advocate position for the duration of the Collective Agreement.
- 4. The Women's Advocates will be afforded two (2) hours per week to perform her duties. This time will be mutually agreed upon between the Women's Advocate and her manager in advance. This time will be scheduled to ensure that it has the least impact to operation.
- 5. The Company agrees to provide the Women's Advocate with access to a confidential phone line to be used exclusively for the duties of the Women's

Advocate. When a female member requests a meeting in person with the Women's Advocate at a VIA location, the Corporation will make its best effort to provide access to a private office so that confidentiality can be maintained when a female employee is meeting with the Women's Advocate.

- 6. The Corporation and the union will develop appropriate communications to inform employees about the advocacy role of the Women's Advocate and provide the contact number for the confidential phone line.
- 7. The Women's Advocate will participate in an initial forty (40) hours training program organized by the Union and annually a three (3) day training program including travel time. The Corporation may select a Corporate representative to participate in the training (This is done separately for employer representatives).
- 8. The Women's Advocates may be called on to assist with other Corporate Programs including Mental Health and Wellness.
- 9. This program is implemented for the duration of the current collective agreement.

If you are agreeable to the above, please indicate your concurrence below.

Agreed on: December 3, 2020\_\_\_\_\_

**Karine Chapados** Specialist Advisor, Employee Relations VIA Rail Canada

## Letter regarding renewal of all appendices

Ms Karine Chapados Specialist Advisor, Employee Relations VIA Rail Canada, Inc. 3 Place Ville-Marie, Suite 500 Montréal, Québec H3B 2C9

Ms. Chapados:

This letter serves to acknowledge that the parties agree to the renewal of all appendices of collective agreements one and two at the completion of the bargaining process.

Respectfully

Dave Kissack President, Unifor Council 4000

## Agreement – System Health and Safety Legislative Representative

VIA RAIL INC. (Herein referred to as the "Corporation")

## AND

## UNIFOR, COUNCIL 4000

(Herein referred to as the "Union")

**WHEREAS** both parties wish to join forces to ensure a safe and healthy workplace for their employees and members;

**WHEREAS** both parties recognize the importance of having a dedicated health and safety representative;

**WHEREAS** both parties recognize that rules of governance must be established in order to supervise the health and safety representative;

**NOW, THEREFORE, THE PARTIES** have reached an agreement on the terms and conditions regarding the health and safety representative.

## IN CONSIDERATION OF THE FOREGOING, THE PARTIES AGREE TO THE FOLLOWING:

## 1.Preamble

The preamble is part of this letter of agreement.

## 2.Nomination

It is the Union's responsibility to nominate an employee as the health and safety representative. This person must reflect the values of the organization and must have an impeccable health and safety record.

The person chosen must be bilingual and able to travel on regular basis. The Employer reserves the right to refuse an employee's nomination for any valid reason, such as known behavioural problems or absenteeism.

The appointment will be for a period of the life of the collective agreement

The appointment must be confirmed through an official letter from the Union to the attention of VIA Rail Canada Inc Labour Relations Director.

#### 3. Seniority and vacation

The health and safety representative will continue to pay union dues and accumulate seniority and will retain seniority rights.

The representative should direct vacation requests to the Director responsible for the Health & Safety function.

### 4.Work schedule

The health and safety representative must be available from Monday to Friday and exceptionally on evenings and weekends for urgent safety matters. He must be physically present at the Employer's facilities during work hours. The representative will be able to work from home as per the employer's policy.

The representative must complete his timesheet and submit it to the Director responsible for the Health & Safety function for validation every week.

### 5.Reporting structure

The manager for the Health & Safety function will exercise functional authority. However, health and safety priorities will be determined by both parties.

#### 6.Equipment

The health and safety representative will be provided with an iPhone and a laptop computer. He/she will be provided access, where available, to an office with filing cabinets, a desk, chairs, a telephone, a photocopy machine, and a fax machine. He/she will be provided access to the Corporation's systems or any system the Corporation uses to track health and safety. Equipment must be returned at the end of the term.

#### 7. Travel expenses

All travel expenses will be approved and shared by both parties.

#### 8.Rate of pay

The employee will be paid at SM Corridor's hourly rate.

## 9. Role and expectations of the health and safety representative

Created under the terms of the National Safety, Health and Environment Agreement signed on February 22, 1999, and nominated by the Union, the Unifor System Health & Safety Legislative Representative supports the Corporation and Unifor achieve their common goal of creating and maintaining a safe and healthy workplace.

The representative works with the Health & Safety department assisting in the implementation of the Health & Safety plan and playing a proactive role in health and safety prevention, always taking into consideration the risk management principles for prioritization. He/she acts as a coach and intervenes with employees and members in order to strengthen the organization's health and safety culture and contributes to ensure a safe and healthy workplace for the employees and members.

The Health and Safety Representative shall perform the duties set out in his or her job description.

This agreement may not be used by any of the parties as a reference for a settlement of any other grievances or arbitration hearing.

Please indicate your concurrence by signing below and returning one (1) duly signed copy for our files.

## Agreement regarding Locomotive Attendant

Friday, April 18, 2022

Mr. David Kissack President - Unifor National Council 4000

Dear Mr. Kissack

Pursuant to *our* recent discussion during this round of bargaining, this letter *will* serve as confirming your proposal for increasing the wages for the locomotive attendant:

"For 2022, in addition to any negotiated wage increases Locomotive Attendants *shall* receive a onetime wage adjustment of \$2/ hour retroactive to January 1, 2022, for *all* hours worked, including overtime.

Retroactivity will be paid no later than 90 days following ratification. Inactive employees will receive their increase upon their return to work."

## Letter of Understanding

#### Between:

#### VIA RAIL INC. (Herein referred to as the "Corporation")

#### AND

#### UNIFOR, COUNCIL 4000 (Herein referred to as the "Union")

## Subject: Pensionable service amendments for certain Employees participating in the Job-Sharing Program

- **WHEREAS** Employees previously and currently participating in the Job-Sharing program have had and have their pensionable service pro rated while participating in the program.
- WHEREAS The parties recognize that a recent Supreme Court of Canada decision has found that failure to allow a certain category of female employees (the "group") who opt into the Job-Sharing Program to buy back their full pensionable service while employed in the Program can constitute unlawful discrimination based on sex;
- WHEREAS The parties wish to take the following measures and enter into the present letter of Understanding to avoid any charge of discrimination be any employee in the group in the future or generally any charge against the Job-Sharing Program or the Pension Plan on this basis;

#### The parties agree to the following:

- 1. The preamble of the present Letter of Understanding forms an integral part thereof;
- 2. Going forward, all employees in the group who join as of the date of the signing of the present letter of agreement the Job-Sharing Program to balance their work and family obligations after a maternity and/or parental leave may opt to be credited with full Cumulative Compensated service for the purpose of determining pensionable service time and in such case be required to make contributions to their pension to reflect such in accordance with the pension guidelines and according to the work schedule prior they join the job sharing program.

- 3. All employees affected by paragraph 2 of this letter of agreement will be required to contact Shared Services before the end of their Job-Sharing program to confirm if they will opt to buy back their pension deficiencies under the methods defined in the Pension Plan for Unionized Employees of VIA Rail Canada inc. An employee must provide proof of caregiver responsibility when reasonably practicable to do so or through a sworn attestation to request a pension buy back.
- 4. Employees wishing to participate in a job-sharing program will be advised of the change when applying for the program.

This letter of understanding may be amended through mutual consent between the parties in the event of a legislative requirement to do so or as a result of an award by a third party of competent jurisdiction which mandates consideration of such actions.

## Letter of Understanding

#### Between:

## VIA RAIL INC. (Herein referred to as the "Corporation")

#### AND

### UNIFOR, COUNCIL 4000 (Herein referred to as the "Union")

## Subject: Wage scale – Reclassification internet agent at the VCC

**WHEREAS** the current negotiation, the Union has expressed the need for a job evaluation for the internet agent position at the VCC.

## THE PARTIES AGREE TO THE FOLLOWING:

- 1. The Corporation is committed to bring this position to the job evaluation committee to determine if there should be an increase for this position.
- 2. Should the Union disagree with the proposed wage rate for the position they may file a grievance.
- 3. The parties mutually agree to resolve the grievance at Med Arb.

### Letter of Agreement

#### Between:

## VIA RAIL INC. (Herein referred to as the "Corporation")

#### AND

## UNIFOR, COUNCIL 4000

(Herein referred to as the "Union")

During the 2021-2022 round of collective bargaining, the Union raised concerns regarding the challenges some Agreement 1 employees had faced in attempting to access or displace onto positions that they have previously been trained and qualified to work.

In understanding these concerns, the parties agree that should an employee exercise their seniority rights through displacement under Article 13 of Agreement 1 to a position they were previously deemed qualified for, but have not worked during an extended period and their ability to perform the essential duties of the position may be in question, the following conditions will apply:

- 1. The employee will tentatively be awarded the position.
- 2. Management will meet with the employee and their Union representative to outline their expectations of the position.
- 3. A familiarization period of not less than five (5) working days will be provided and where upon request, the employee will be provided with 'on-the-job' mentoring, and should the employee properly demonstrate their qualification, they will be awarded the position. During this period the previous incumbent will remain on the position.
- 4. After the familiarization period, should the employee be unable to demonstrate their ability to perform the essential duties of the position, they will be removed from the position and Article 13.7 of the collective agreement shall apply.
- 5. Should the employee be unable to demonstrate qualification after the familiarization period, they shall be eligible to qualify for the position at the next available training sessions provided under Article 16.4 of the collective agreement.

This policy will not apply to the position of Locomotive Attendant, whereas the employer will be mandated to inform qualified employees that their rules qualifications will be lapsing and will be required to provide sufficient training to maintain qualification.

## Letter of Understanding

#### Between:

VIA RAIL INC. (Herein referred to as the "Corporation")

#### AND

## UNIFOR, COUNCIL 4000 and Local 100

(Herein referred to as the "Unions")

### Subject: Racial Justice Advocate Program and Anti-Racism Action Plan at VIA Rail

Dear Sirs,

This refers to our recent discussions at the bargaining table pertaining to unique situations faced by Black, Indigenous, or racialized community in the workplace.

Establishment of a Racial Justice Advocate

In recognition of societal racism, the Parties agree to identify a Racial Justice Advocate who will ast as an ambassador in VIA Rail's Diversity and Inclusion program.

A Racial Justice Advocate is an individual who identifies as a member of the Black, Indigenous or racialized community.

The Unifor Local/Council Union President and VIA Rail representative will be responsible for the selection of the Racial Justice Advocate.

The Racial Justice Advocate is a workplace representative who will participate in the VIA Rail Diversity and Inclusion program.

The activities will be performed in relation with VIA Rail Diversity and Inclusion program. Support will also be provided by the VIA Rail Diversity and Inclusion program.

If you are agreeable to the above, please indicate your concurrence below.

Agreed on: July 12<sup>th</sup>, 2022